





Assessment of Concrete Structures Including Corrosion and Cracks

MATTIAS BLOMFORS

Department of Architecture and Civil Engineering CHALMERS UNIVERSITY OF TECHNOLOGY Gothenburg, Sweden 2020

THESIS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

Assessment of Concrete Structures Including Corrosion and Cracks

MATTIAS BLOMFORS

Department of Architecture and Civil Engineering Division of Structural Engineering Concrete Structures CHALMERS UNIVERSITY OF TECHNOLOGY Gothenburg, Sweden 2020 Assessment of Concrete Structures Including Corrosion and Cracks

MATTIAS BLOMFORS ISBN: 978-91-7905-340-6

© MATTIAS BLOMFORS, 2020

Doktorsavhandlingar vid Chalmers tekniska högskola New series number: 4807 ISSN 0346-718X

Department of Architecture and Civil Engineering Division of Structural Engineering Concrete Structures Chalmers University of Technology SE-412 96 Gothenburg Sweden Telephone: +46 (0) 31 772 1000

Cover:

Illustrations of aspects treated in this thesis, such as uncertainty, modelling level, cracks and corrosion, are shown in the foreground. In the background a photograph of an example reinforced concrete bridge located in the Miraflores district of Lima, Peru, is shown.

Chalmers Reproservice Gothenburg, Sweden 2020

To Lova and Sara

Assessment of Concrete Structures Including Corrosion and Cracks

MATTIAS BLOMFORS Department of Architecture and Civil Engineering Division of Structural Engineering, Concrete Structures Chalmers University of Technology

Abstract

Reinforced concrete (RC) structures constitute a major proportion of the built environment and society relies continuously on their service. Many of these structures were built in the era following the Second World War and are thus approaching the end of their intended service life. The likelihood of deterioration increases with time and so damage caused by, say, corrosion is not uncommon. Also, increased demands are often laid on the load-carrying capacity of existing bridges, aimed at increasing utilisation of the road network by allowing heavier vehicles. Simply dismantling and re-constructing all bridges at the end of their designed service life, or taking needless strengthening measures, is unsustainable. Rather, improved methods of assessing the capacity of existing infrastructure are needed.

The current work has aimed to develop improved, reliable assessment methods. Its focus areas were structures with reinforcement corrosion and structures with cracks from previous loading. Both simplified and advanced methods of evaluating anchorage capacity were developed for concrete structures with corroded reinforcement. The simplified method modifies the bond stress-slip relationship and is calibrated against a large database of bond tests, with the safety margin ensured by deriving partial safety factors. The advanced method is based on finite element (FE) analysis, with tensile material properties altered for elements positioned at the splitting cracks along the reinforcement. The latter method was also investigated for RC without corrosion damage but with cracks from previous loading. Design results from advanced nonlinear FE analyses (meaning results with a proper safety margin) are obtained by applying a "safety format". The current work investigated whether safety formats available in *fib* Model Code 2010 also ensured reliable design capacities for structures with somewhat complicated load application and geometry; in this case, a concrete frame subjected to vertical and horizontal loads.

The results indicate that the anchorage capacity may be reasonably well estimated by using the simplified method. The proposed partial safety factors also provided sufficient safety margin. Furthermore, in the advanced anchorage assessment, the capacity could be estimated solely from weakened tensile properties located at the position of the splitting cracks and without input concerning the corrosion level. Moreover, by including cracks from previous loading in advanced modelling, improved predictions of the failure mode, ultimate capacity and ductility were demonstrated. Lastly, in the investigation of safety formats for nonlinear FE analysis, the method of *estimating a coefficient of variance of resistance* (ECOV), did not reach the intended safety level. However, the *global resistance factor method* (GRF) and *partial factor method* (PSF) did.

This work has the potential to improve both simplified and advanced assessment methods, providing more sustainable infrastructure management in the future.

Keywords: reinforced concrete, corrosion, reinforcement bond, nonlinear FE analysis, cracks, assessment

CHALMERS, Architecture and Civil Engineering

Assessment of Concrete Structures Including Corrosion and Cracks

MATTIAS BLOMFORS Arkitektur och samhällsbyggnad Konstruktionsteknik, Betongbyggnad Chalmers tekniska högskola

Sammanfattning

Betongkonstruktioner utgör en stor del av bebyggelsen i den industrialiserade världen och samhället förlitar sig ständigt på deras funktion. Många av dem byggdes under återhämtningen efter andra världskriget och börjar därför närma sig slutet på sin tekniska livslängd. Sannolikheten för skador ökar också över tid, så exempelvis korrosionsskador är inte ovanligt. Vidare efterfrågas med tiden generellt sett ökad bärförmåga hos befintliga broar, så att tyngre fordon kan användas och därigenom öka vägnätets resursutnyttjande. Att helt sonika riva äldre broar vid slutet av deras tekniska livslängd och bygga nya är varken miljömässigt eller ekonomiskt hållbart. Istället behövs bättre metoder för att bedöma kapaciteten hos befintliga betongkonstruktioner.

Målet med denna avhandling var att utveckla metoder för tillförlitliga tillståndsbedömningar. Arbetet fokuserades på konstruktioner med armeringskorrosion och på konstruktioner med sprickbildning på grund av tidigare belastning. För betongkonstruktioner med rostande armering utvecklades både förenklade och avancerade metoder för att bedöma förankringskapaciteten. Den förenklade metoden innebar förändrad vidhäftning som funktion av glidning och kalibrerades mot en stor testdatabas. Vidare tryggades säkerhetsmarginalen hos beräkningsresultaten genom att säkerhetsfaktorer för modellen togs fram. I den avancerade metoden, som baseras på finita element (FE) analys, försvagades betongens dragegenskaper hos de element som innehöll de spjälksprickor som den korroderade armeringen orsakat. Den avancerade metoden användes också för betongbalkar utan korrosionsskador, men med sprickor på grund av tidigare belastning. För att erhålla resultat i enlighet med specificerad säkerhetsnivå från icke-linjära FE analyser används så kallade säkerhetsformat. I detta arbete undersöktes om säkerhetsformaten som finns tillgängliga i *fib* Model Code 2010 ger önskad säkerhetsnivå även för konstruktioner med tämligen komplicerad geometri och belastning, närmare bestämt en betongram utsatt för vertikal och horisontell last.

Resultaten visar att försäkringskapaciteten kan uppskattas med hjälp av förenklade metoder. Vidare gav de framtagna säkerhetsfaktorerna tillräcklig säkerhetsnivå i resultaten. Med den avancerade metoden kunde förankringskapaciteten uppskattas enbart genom försvagade element vid spjälksprickornas positioner, utan att använda korrosionsnivån i analysen. Dessutom, genom att sprickor på grund av tidigare belastning beaktades i den avancerade modellen, förbättrades bedömningar av brottmod, kapacitet och seghet. Slutligen, ett av säkerhetsformaten (förkortat ECOV) för icke-linjär FE analys gav för låg säkerhetsnivå medan de två andra gav tillfredsställande nivå (förkortade GRF och PSF)

Arbetet har möjliggjort förbättringar av både förenklade och avancerade metoder för tillståndsbedömning och tagit ett steg mot mer hållbar infrastrukturförvaltning i framtiden.

Nyckelord: armerad betong, korrosion, vidhäftning, icke-linjär FE analys, sprickor, tillståndsbedömning

Table of contents

Abstract							
Sa	SammanfattningII						
Та	able of	cont	tents	III			
Pı	eface.			. V			
Li	st of p	ublic	cations	VII			
1	Intr	oduc	tion	1			
	1.1	Bac	kground	1			
	1.2	Ain	n of the research	3			
	1.3 Methodology		hodology	3			
	1.4	Lim	itations	3			
	1.5	Orig	ginal features	4			
	1.6	Out	line of the thesis	4			
2	Ove	erviev	w of structural assessment	5			
3	Stru	ictura	al analysis of damaged structures	9			
	3.1	Fini	te element analysis of reinforced concrete	9			
	3.1.	1	Linear finite element analysis	10			
	3.1.	2	Nonlinear finite element analysis	10			
3.1		3	Strategy for conducting finite element analysis	10			
	3.2	Dan	nage caused by corrosion	12			
	3.2.	1	Corrosion of reinforcement in concrete	12			
	3.2.	2	Structural effects of corrosion damage	13			
3.3		Moo	delling of corrosion-damaged structures	15			
	3.4	Cra	cking in reinforced concrete	17			
	3.5	Inco	prporating pre-existing cracks in FE modelling	19			
4	Bac	kgro	und on structural reliability assessment	23			
	4.1	Prin	ciples of limit state design	23			
	4.2	Unc	ertainty of basic variables	24			
	4.3	Targ	get reliability level	25			
	4.4	Mea	asures of structural reliability	26			
	4.5	Stru	ctural reliability methods	28			
	4.5.	1	First-order reliability method	29			

	4.5.	2 Monte Carlo simulations)			
	4.5.	3 Response surfaces	2			
5	Sun	nmary of appended papers	3			
6	Con	clusions and future research	1			
	6.1	Conclusions	1			
	6.2	Suggestions for future research	5			
References						

APPENDED PAPERS

Paper I: Evaluation of safety formats for non-linear finite element analyses of statically indeterminate concrete structures subjected to different load paths

Paper II: Engineering bond model for corroded reinforcement

Paper III: Partial safety factors for the anchorage capacity of corroded reinforcement bars in concrete

Paper IV: Incorporation of pre-existing longitudinal cracks in finite element analyses of corroded reinforced concrete beams failing in anchorage

Paper V: Incorporation of pre-existing cracks in finite element analyses of reinforced concrete beams prone to shear failure

BIBLIOGRAPHY, CONCRETE STRUCTURES

Doctoral Theses

Licentiate Theses

Preface

The work presented in this thesis was conducted mainly within two research projects. The first was conducted between October 2015 and September 2018, at the Division of Structural Engineering at Chalmers University of Technology in Gothenburg and the CBI Swedish Cement and Concrete Research Institute in Borås, Sweden. This research project was funded by the Swedish Road Administration, CBI Swedish Cement and Concrete Research Institute's A-consortium and SBUF. The work on the second research project was conducted between October 2018 and September 2020, at Chalmers University of Technology and with financial support from FORMAS.

Firstly, I would like to thank my supervisor Assoc. Prof. Kamyab Zandi for his continuous feedback, his positivity and his responsiveness to my wishes throughout the years. I also want to thank my co-supervisor and examiner, Prof. Karin Lundgren for providing countless valuable ideas and comments. Her extraordinary ability to bring clarity and reassurance in times of need has made this journey much more enjoyable. Thanks also to Oskar Larsson Ivanov who, in his role as co-supervisor for the first part of this work, provided much-appreciated input. I would also like to acknowledge Dániel Honfí and Per Kettil, for their good ideas and comments. And I would like to thank all reference group members, in both research projects, for sharing their knowledge and helping guide the work. Lastly, I would like to thank Morten Engen, for being a source of inspiration and awakening my interest in research six years ago.

To my current and former colleagues at the Division of Structural Engineering, thanks for creating an open and friendly environment and for sharing your thoughts and experiences. Carlos Gil Berrocal, I'm glad to have worked more closely with you this last year. Thanks for sharing your knowledge, experience and humour. Ignasi Fernandez Perez, our lives have changed quite a lot since we went racing with our bikes, but I think you'll agree that it's mostly for the better. Adam Ścięgaj, your sharp mind and sense of humour makes every discussion interesting. Samanta Robuschi, it has been a pleasure to get to know the most untraditional Italian and observe the culinary progress you have made. Sebastian Almfeldt, sincere thanks for your help during the experimental part of this work. You always took the time to explain things in an informative manner, with one or two jokes along the way.

This work would not have been possible without the support of my family; in particular from my beautiful wife, Sara. I'm so happy we found each other; we are so different but at the same time so very much alike. Thank you for showing me what is truly important in life. My beloved daughter Lova, you effectively prevented me from worrying too much about work, when at home. Stories, dance and laughter filled our evenings instead. Oh, I am so happy I've got you!

Mithadunt

Mattias Blomfors Gothenburg, 2020

List of publications

This compilation thesis consists of an introductory part and five appended publications, specifically:

Journal Papers

- I. Blomfors, M., Engen, M., Plos, M. (2016), Evaluation of safety formats for nonlinear finite element analyses of statically indeterminate concrete structures subjected to different load paths, Structural Concrete, 17: 44-51. doi: 10.1002/suco.201500059.
- II. Blomfors M., Zandi K., Lundgren K., Coronelli D. (2018), Engineering bond model for corroded reinforcement, Engineering Structures, 156: 394-410. doi: 10.1016/j.engstruct.2017.11.030.
- III. Blomfors M., Larsson Ivanov O., Honfi D., Engen M. (2019), Partial safety factors for the anchorage capacity of corroded reinforcement bars in concrete, Engineering Structures, 181: 579-588. doi: 10.1016/j.engstruct.2018.12.011.
- IV. Blomfors M., Lundgren K., Zandi K. (2020), Incorporation of pre-existing longitudinal cracks in finite element analyses of corroded reinforced concrete beams failing in anchorage, Structure and Infrastructure Engineering. doi: 10.1080/15732479.2020.1782444.
- V. Blomfors M., Berrocal C.G., Lundgren K., Zandi K. (2020), Incorporation of preexisting cracks in finite element analyses of reinforced concrete beams prone to shear failure. *Submitted for publication*.

AUTHOR'S CONTRIBUTION TO JOINTLY PUBLISHED PAPERS

The appended papers were prepared in collaboration with co-authors. The contribution of the author of this doctoral thesis to the appended papers is described below.

In **Paper I**, the author was partly responsible for planning the study. He conducted the literature review, conducted the non-linear finite element analyses and applied the reliability methods. The author also took the lead in writing the paper and the co-authors assisted in planning the study, discussing the results and writing the paper.

In **Paper II**, the author planned the main part of the paper, developed the bond model and took the lead in writing the paper. The co-authors participated in discussing the model development and results and in writing the paper.

In **Paper III**, the author planned the main part of the paper, derived the partial safety factors and conducted the reliability analyses. The author also took the lead in writing the paper. The co-authors participated in discussing the methods and results and in writing the paper.

In **Paper IV**, the author planned the main part of the paper, developed the methods for implementing cracks and conducted the finite element analyses. The author also took the lead in writing the paper. The co-authors participated in the discussion of the methods and results and in writing the paper.

In **Paper V**, the author was, with Berrocal, responsible for planning and executing the experimental study. The author planned the main part of the paper, further developed the methods for implementing the cracks and conducted the finite element analyses. The author also took the lead in writing the paper. The co-authors participated in the discussion of the methods and results and in writing the paper.

OTHER PUBLICATIONS RELATED TO THIS THESIS:

In addition to the appended papers, the author of this thesis has also contributed to the following publications.

Licentiate Thesis

Blomfors M. (2017). *Reliable Assessments of Concrete Structures with Corroded Reinforcement: An Engineering Approach*. Chalmers University of Technology, Licentiate Thesis, Gothenburg, ISSN 1652-9146.

Conference Papers

- C-I. Blomfors M., Zandi K., Lundgren K., Larsson O., Honfi D. (2016): Engineering Assessment Method for Anchorage in Corroded Reinforced Concrete, in 19th IABSE Congress Stockholm, Sweden 21-23 September 2016, pp. 2109-2116.
- C-II. Blomfors M., Zandi K., Lundgren K. (2016): Development of engineering assessment method for anchorage in reinforced concrete, in *Nordic Concrete Research*, 2/2016, pp. 63–78.
- C-III. Blomfors M., Zandi K., Lundgren K., Larsson O., Honfi D. (2017): Reliable Engineering Assessments of Corroded Concrete Structures, in XXIII Nordic Concrete Research Symposium, Aalborg, Denmark, 21st-23rd August 2017, pp. 245-248
- C-IV. Blomfors M., Larsson O., Honfi D., Zandi K., Lundgren K. (2018): Reliability analysis of corroded reinforced concrete beam with regards to anchorage failure, in 6th International Symposium on Life-Cycle Civil Engineering, Ghent, Belgium, 28th-31st October 2018, pp. 337-344.
- C-V. Blomfors M., Zandi K., Lundgren K. (2019): Incorporation of cracks in finite element modelling of existing concrete structures, in *12th International Workshop on Structural Health Monitoring, Stanford, USA, 10th-12th September 2019, pp. 1611-1618.*

Popular Science Papers

- P-I. Tahershamsi M., Blomfors M., Fernandez I., Lundgren K., Zandi K. (2016): Modellering av förankringskapaciteten i betongkonstruktioner med rostande armering, Bygg & Teknik, 7/16, pp. 16-20.
- P-II. Blomfors M., Lundgren K., Zandi K. (2019): Digitala tvillingmodeller underlättar bedömning av risker, Samhällsbyggaren, 4/19, pp. 24-25.

1 Introduction

1.1 Background

Reinforced concrete (RC) structures constitute a large part of the built environment and society continuously relies on their service. Core societal features such as transportation, shelter and electricity depend on well-functioning concrete structures such as bridges, houses, dams, and so on. Many concrete structures were built during the economic expansion following the Second World War [1] and are now approaching the end of their designed service life. Enabling a safe service-life extension of those structures, or accommodating increased loads often requires an assessment of capacity [2]. As long-term exposure to the environment increases the likelihood of deterioration, the structural models used in these assessments must be able to account for potential deterioration. Reinforcement corrosion is the most common type of deterioration in concrete structures [3]. Advanced corrosion may result in considerable loss of reinforcement bar cross-section and reduced bond properties due to loss of confinement and rib area, plus the weak frictional properties of the corrosion products [4]. Unlike the ductile failure modes typically sought during design, loss of bond might lead to an abrupt failure [5], meaning that users of the structure are unable to avoid imminent danger. Thus, bond capacity may be a major concern when assessing corroded structures. This phenomenon has been studied in recent decades, cf. [6-10] and advanced models have been developed to simulate the phenomenon, cf. [11-13]. Nevertheless, when practical assessments of corrosion-damaged structures are conducted, advanced methods are typically unfeasible due to the large amount of time they take. Simply disregarding the contribution of corroded reinforcement to capacity is a quick and practical approach but may lead to overly conservative estimates of structural capacity. Thus, a need has been identified for an engineering method of anchorage assessment. To enable practical use of the results obtained from such a method, they must be able to provide a sufficient safety margin (or reliability) [14]. To this end, "safety formats" are used to ensure that the failure probability is less than acceptable. Semi-probabilistic safety formats in the form of partial safety factors are most commonly used [15]. For a model to assess the anchorage of corroded reinforcement (and with professional engineers as the intended users), the author believes the well-known partial safety factor approach to be preferable in ensuring safety.

The use of simplified models of anchorage assessment of corroded bars is a step forward from overly conservative assumptions. There are also other types of structural analyses (of, say, shear capacity) for which increased complexity may be justified. The complexity of structural analyses varies, from simple hand calculations to detailed numerical analyses, often conducted by nonlinear finite element (NLFE) analysis. The underlying principle is that, as the complexity of the analysis increases, so does the accuracy of the estimate and the effort from the analyst [16]. In other words, a structure's capacity might be more accurately assessed if a more elaborate structural analysis is used. However, this increases demand on the analyst's competence, time and resources. In structural analyses, complex methods (such as NLFE analyses) are typically used to capture realistic behaviour of a structure. When conducting an NLFE analysis, choices need to be made concerning force equilibrium, kinematic compatibility and constitutive relations [17]. These choices influence the "modelling uncertainty" (accuracy of prediction), which have been studied in [18, 19] for example. Guidelines on how to conduct NLFE analyses of concrete structures have also been published [20]. The author regards such

recommendations as an important step towards more consistent, and hopefully also decreased, modelling uncertainty. Furthermore, since NLFE analyses evaluate capacity on the global level, safety formats based on global resistance have been put forward which provide a design resistance with a sufficient safety margin, cf. [16, 21]. Since the modelling uncertainty needs to be considered when determining the design resistance, an underestimate may lead to insufficient reliability. However, concerning the NLFE safety formats suggested by *fib* [16], it is unclear how the loading sequence should be addressed. It is, therefore, of interest to investigate whether different choices regarding the loading sequence may influence the resulting design resistance, particularly for more complex loading and geometry.

In assessing structures, their behaviour at the time of the assessment should be evaluated [22]. Furthermore, cracking in RC structures is common and not necessarily undesirable, as cracking is needed if mild reinforcement steel is to carry any significant tensile force. The causes of cracking in concrete members are numerous. A non-exhaustive list includes (apart from cracking induced by reinforcement corrosion) cracking due to other types of internal and external restraint and due to live loads acting on the structure [23]. Design codes stipulate crack control measures that meet durability and aesthetic requirements, cf. [24, 25]. However, cracking may also affect the ductility and structural capacity of an RC member [26], thus posing a risk to the integrity of the structure. Although the design codes are followed, some cracks may grow past the specified limits. Therefore, there is a need for improved assessment methods for reinforced concrete infrastructure. that capable of incorporating cracks formed in the structure at the time of are assessment. These are referred to as pre-existing cracks since they are present at the start of the NLFE analysis (as opposed to cracks forming during the analysis). One way of facilitating an assessment with updated structural information is to use Digital Twin (DT) models [27]. A DT is a virtual copy of the structure in which information collected during its service-life (through, say, different types of sensors) is stored. In a proposal for a DT of civil infrastructure [28], FE analysis provides insights on the structural capacity based on updated information about the structure. The latest advancements in techniques of data collection, such as the application of inspection and monitoring drones [29-31] and fibreoptic sensing [32, 33], allow for crack data to be collected with unprecedented accuracy and frequency. Therefore, to leverage information on pre-existing cracks, the FE simulations used in structural assessments need to include this information.

There is a need for improved assessment methods, to make better use of current concrete infrastructure and avoid unnecessary repair and decommissioning. Some avenues for improving current assessment methods are pursued in this thesis.

1.2 Aim of the research

This research aimed to develop reliable structural assessment methods, for concrete structures with corrosion damage and pre-existing crack patterns. Accordingly, the following questions were formulated and addressed:

- might available safety formats for nonlinear analyses also be used for structures with more complicated global failure mechanisms than a simple concrete beam? For example, an indeterminate structure subjected to non-proportional loading;
- is it possible to obtain representative estimates of the anchorage capacity of corroded reinforcement using a simplified model that is suitable for an engineering context?
- might such a model be used in conjunction with a semi-probabilistic safety format to ensure a sufficient safety margin?
- might the influence of cracks on the structural behaviour be represented without modelling their formation (load history)? Specifically:
 - o the effect of corrosion-induced splitting cracks (along the longitudinal reinforcement) on the anchorage capacity;
 - o the effect of cracks (formed under previous external loading) on the moment and shear capacity.

1.3 Methodology

The above research questions were answered by completing the following tasks, listed in the same order:

- investigate the applicability of various safety formats to a statically indeterminate frame, subjected to vertical and horizontal loading;
- develop and validate a model for anchorage assessment of corroded reinforcement, intended for use in practical applications;
- equip the proposed model for anchorage assessment with a semi-probabilistic safety format, to ensure a proper safety margin for the results;
- investigate methods of incorporating corrosion-induced longitudinal cracks in structural analysis;
- explore the extendibility of a selection of the aforementioned methods, to incorporate cracks due to previous external loading.

1.4 Limitations

A summary of the main limitations of the present work is presented below:

- the evaluation of safety formats in **Paper I** considered one type of structure with a certain load configuration. An exhaustive evaluation would also consider other types of structures and, say, other types of loading;
- only failures due to loss of anchorage were considered for corroded structures. In other words, reduction of reinforcement cross-section leading to other failure modes was not studied in **Papers II-IV**;
- in Paper II, the calculation model for anchorage was calibrated based on experimental tests of corroded specimens. These specimens, and those used for validation in Paper IV, were artificially corroded. Although the transferability to natural corrosion is

unclear, artificial corrosion induced by relatively low-current densities was considered acceptable due to the lack of experimental results with natural corrosion;

- in **Paper III**, when calibrating the partial safety factors and verifying the reliability levels, no time-dependence of the resistance due to damage propagation was considered. A reference period of one year was chosen, to reduce the influence of this simplification;
- the incorporation of cracks in **Paper V** only considered cracks due to previous external loading. Since this thesis also includes work with corrosion damage, it is important to clarify that, in the case of corrosion-induced cracks, the corrosion level (and possibly also the distribution and shape of the pits) needs to be accounted for when assessing bending and shear failure modes. This was not addressed in the current work.

1.5 Original features

The original features of the present work may be summarised as follows:

- a bond model, based on a 1D bond stress-slip relationship, was developed and calibrated for corroded reinforcement bars in concrete;
- partial safety factors were derived for the above model;
- methods for incorporating pre-existing cracks into NLFE analysis were developed, based on the visually obtained crack pattern.

1.6 Outline of the thesis

This thesis consists of an introductory part and five appended papers. The former is divided into six sections:

Section 1 gives a background to the work, and presents its aims, methods and limitations, plus its original features.

Section 2 presents an overview of structural assessments.

Section 3 describes the structural consequences of damage from reinforcement corrosion and also from cracks induced by previous loading.

Section 4 provides an overview of structural reliability analyses.

Section 5 briefly summarises the appended papers.

Section 6 states the main conclusion drawn from this work and puts forward suggestions for future research.

2 Overview of structural assessment

This thesis deals with structural assessments of RC structures. An overview is given below and describes where the proposed models and methods fit into the framework of assessment. It is based on [22] which, in turn, is based on [14, 34 and 35].

The design of new structures and assessment of existing ones are fundamentally different, in that the latter has already been realised. At the design stage of a building process, many circumstances are unknown and will be determined in the future. For example, any differences between the original and the as-built design, or any structural damage sustained during the service life of the structure. However, for an existing structure, this information is known (to some extent at least). The original design of the structure, the as-built design, or both, may have been subjected to changes. The structure may also have been damaged due to misuse or deterioration (with reinforcement corrosion being the most common example [36] for concrete structures). A proper structural assessment will use the current structural state and loading conditions.

The safety of a structure is typically assessed for one (or more) of the following reasons:

- changes in use: to demonstrate the safety of a structure if the conditions of use (and thereby the associated loads) have changed;
- addition of new structural members: to study the consequences for structural behaviour if new elements are added to the structural system;
- in the case of repair: to determine suitable repair measures. These may differ depending on the damage (caused by, say, accidents, natural phenomena or environmental impacts). The safety level of a repaired structure is also of interest;
- doubts as to safety: to study the structural safety relating to concerns arising due to other reasons;
- other circumstances: to accommodate any requirements by insurance companies, official bodies or owners.

Two main principles are generally followed in reliability assessments:

- use of current codes: the codes valid at the time of assessment should be used. Earlier codes, such as those valid at the time of the structure's design, should serve only as guidance;
- use of actual structure: the actual geometries and actual applied loads plus in-situ material properties should be used. Given that the structural behaviour at the time of assessment should be estimated, design documentation should only be used for guidance.

An overview of the processes relating to structural reliability assessment is given in Figure 2.1.



Figure 2.1. Overview of the processes related to structural reliability assessment of an existing structure, adopted from [14].

Not all parts of a structure have to be included in all assessments. Some parts may be excluded if unaffected by changes (due to repair or change of use, for example) and are neither damaged nor suspected of being insufficiently reliable. The processes involved in structural reliability assessment are illustrated in a flow chart in Figure 2.2.

At the beginning of an assessment process, the objectives (in terms of the structure's future performance) should be set by the client, the assessing engineer and any relevant authorities. In the next step, likely scenarios associated with changes to structural conditions or actions should be specified, to help identify any critical situations for the structure. The assessment and any interventions to ensure the structure's reliability are based on these identified scenarios.

A preliminary assessment is then started, in which the state of knowledge concerning the structure is established by studying available documents and other material. The structural system and any damage are identified by visual examination during a preliminary on-site inspection. Damage detectible by visible inspection typically comprises deformations, cracks, spalling and signs of corrosion. The damage is graded in qualitative terms (for example "none", "minor", "moderate", "severe", "destructive" or "unknown". It is worth noting that the corrosion level is difficult to quantify using non-destructive measures, as elaborated upon in Section 3.2.1. The information acquired in the preliminary assessment serves as a basis for initial checks to identify current and future deficiencies that are important to the structure's safety and serviceability. If these preliminary checks clearly indicate an unsafe condition for the structure, danger to the public should be mitigated by prescribing immediate measures. The preliminary assessment also forms a basis for determining whether further investigations are necessary. If so, a detailed assessment may be conducted.

A detailed assessment includes an in-depth study of all available documentary information. If concerns are raised regarding the trustworthiness of the information (regarding the structural

dimensions and the material properties, for example), this should be collected instead during the detailed inspection and through material testing. The quantitative results from the detailed inspection will yield updated values for relevant parameters in the structural analysis, plus the actions that have been determined for the structure. As noted, the structural analysis comprises information about the load effects from actions on the structure and the capacity of its structural components. It is, therefore, of the utmost importance that any deterioration of the existing structure is considered in the analysis and that suitable reliability assessment methods are used. **Papers II, IV** and **V** contribute to this area. Structural analysis, including reinforcement corrosion, was conducted in the first two of these papers and pre-existing cracks were included in the third. It is also possible to use non-destructive testing to estimate the load-bearing capacity and certain properties of a structure.

Verification of the structural safety is conducted by ensuring that the structure meets the target reliability level (which was the topic of **Papers I** and **III** in this thesis). The verification may be done using "adjusted partial safety factors", which may consider alternative values for the reliability index, remaining service-life and updated information of, say, material properties [37]. Moreover, the verification basis may comprise the past performance of the structure. The assessment results should be documented in a report, providing conclusions and suggested interventions.



Figure 2.2. Flow chart for structural safety assessments, adopted from [14].

3 Structural analysis of damaged structures

Section 2 described how assessments involve structural analyses of the actual structure. This includes the *as-built* geometry and any damage. Therefore, it is important that the structural analysis methods available to engineers can factor in key damage mechanisms. This work addressed elements of this research area, by proposing both simplified and advanced methods of including the effect of reinforcement corrosion on anchorage capacity (**Papers II** and **IV**). Furthermore, a structural analysis of RC with cracks from previous loading was also addressed (**Paper V**). Since a finite element analysis of reinforced concrete was used in several of the appended papers, an introduction to this topic is given here. There then follows an overview of corrosion damage and cracking in RC and, to conclude, methods for including corrosion damage and cracks from previous loading in structural models.

3.1 Finite element analysis of reinforced concrete

The finite element (FE) method is a technique for solving the field problems used in many areas of engineering and research [38]. These fields typically describe stresses and displacement in the context of structural engineering. However, the FE method is generally applicable in solving various differential equations.

The modelled structure is divided into small pieces (*finite elements*) which are connected together at node points. The arrangement of elements and nodes is called an "FE mesh". Figure 3.1 shows an example of mesh discretisation. The method approximates the field over the finite elements (often by a linear or polynomial distribution), allowing the weak form of the differential equations to be solved. Although FE analysis does not provide an exact solution, the FE mesh may be arranged to make the approximated solution's accuracy sufficient.



Figure 3.1. Example of FE mesh discretisation: a) picture of the Hoover dam [39], b) simplified 3D FE mesh of the dam and c) mesh for an axisymmetric FE model.

Due to cracking, material behaviour and the like, reinforced concrete shows a nonlinear response when subjected to increasing load. Therefore, a nonlinear finite element (NLFE) analysis should be used when a realistic response is sought from an RC structure. However, linear FE (LFE) analyses are commonly used in designing concrete structures. For instructional reasons and for completeness, both linear and nonlinear FE analysis are presented below.

3.1.1 Linear finite element analysis

In LFE analysis, the materials are described by linear elastic relationships and small deformations are assumed [40]. The calculation may be done in one step, as the applied load does not influence the material properties or the geometrical or boundary conditions. A traditional RC design (based on FE modelling) uses global LFE analyses to determine the load effects and nonlinear sectional or regional models to determine the resistance. One valued aspect of engineering practice is the superposition of loads, whereby the load effect from a combination of loads may be found by addition (or superposition) of the individual load effects. This allows for the efficient treatment of load combinations and is especially important in the case of large LFE models checked against multiple load combinations. An appropriate LFE analysis provides a solution in equilibrium with the applied loads and the resulting forces may be used to design reinforcement (given that sufficient ability to redistribute forces is provided). To this end, design guidelines have been proposed, cf. [41].

3.1.2 Nonlinear finite element analysis

To represent the behaviour of an RC structure more realistically, an FE analysis needs to be conducted that includes nonlinear aspects [40]. The dominant sources of nonlinearity in RC structures are the material behaviour plus the geometry and boundary conditions. For example, cracking of the concrete plus yielding and hardening of the reinforcement are examples of nonlinear material behaviour that may be represented by NLFE analysis. Geometric nonlinearities are characterised by deformation of the structure due to the applied loading; the changed geometry then influences the structural resistance (as with second-order moment effects in slender columns, for example). Moreover, a slab supported on the ground may be deemed to have nonlinear boundary conditions. This because the available horizontal reaction forces depend on the magnitude of the vertical load (due to friction).

3.1.3 Strategy for conducting finite element analysis

Several choices need to be made to successfully conduct an FE analysis and evaluate the results. The description here is from [40] but is complemented by additional information concerning the solution strategy from [17]. An overview of a typical strategy for FE analysis is shown in Figure 3.2 and the rubrics are addressed in the sections below. For specific recommendations, the reader is referred to [20]. These recommendations were also largely followed in the FE analyses conducted within this thesis.



Figure 3.2. Overview of a strategy for conducting a finite element analysis, modified from [40].

Structural model

The starting point for choosing a structural model in general, and an FE model in particular, is to establish the desired outcome from an analysis. Clarifying this aids in idealising the structure. For example, if shear failure should be included in the FE analysis of a beam, this will influence the choice of FEs used in discretisation because the elements should be able to reflect shear failure. Based on the desired outcome, the FE model may be idealised using 1D (a beam or truss), 2D (a plane stress/strain element or shell) or 3D (a brick or tetrahedron) FEs, or a combination of these. The model's level of detail and whether certain regions need to be modelled more accurately due to their importance are also guided by well-thought-out analytical goals.

The choices of loading and boundary conditions for the FE model should also be carefully considered; they are simplifications of reality but should still represent important features of the actual structure. In many cases, it is possible to take advantage of the symmetry conditions of the load and geometry and thus reduce the size of the structural model. However, the analyst should be aware of the influence on the global response and remember that the failure mode is often asymmetric in reality (due to, say, local variations in material strength). Additionally, for NLFE analysis, a determination should be made as to which sources of nonlinearity to include (relating to, say, material behaviour, geometry or boundary conditions).

Solution strategy

A solution strategy for FE analysis comprises choices that may be sorted into three groups concerning: i) constitutive relations on the material level, ii) kinematic compatibility on the element level and iii) force equilibrium on the structural level [17]. In a linear elastic model, the constitutive relations are linear, meaning that the stiffness of the material is independent of the stress level. However, for nonlinear material model of, say, concrete, the stiffness response changes markedly with changes in stress level.

Establishing the kinematic compatibility conditions in the model means discretising it into a number of FEs, with associated degrees of freedom. Many FE formulations treat the displacements as unknown, although alternatives exist. Once the FE mesh is constructed, the element stiffness matrices may be derived using the corresponding material stiffness matrices. The global stiffness matrices are obtained by assembling element stiffnesses according to their interconnections.

Resolving the unknowns (here displacements are used as a typical example) using force equilibrium may be done in one step for a linear model. However, this is not possible for a nonlinear analysis, where the stiffness changes with changes in the stress state. An iterative procedure must therefore be used to solve the equilibrium equations. These are typically based on gradually increasing the load or changing nodal positions (often referred to as "load steps"). For each load step, the equilibrium equations are solved iteratively, using the Newton-Raphson method or other related methods [42]. The system of equations is updated based on the sources of nonlinearity (such as nonlinear material behaviour) and solved for a new displacement field. This iterative process is stopped when the pre-set convergence criterion/ criteria are reached. These may be based on such things as normalised values related to energy, out-of-balance force or displacement. Furthermore, the analysis should be stopped if there is a diverging solution. Accordingly, upper bound values may be applied to the same criteria used in checking convergence. If these values are exceeded, the analysis is aborted.

Finally, for some NLFE models, saving all the information generated during the analysis is not feasible due to the large amount of data. For others, it is unnecessary. The outputs of interest should be specified and be in logical agreement with the goal of the analysis.

Post-processing

If the results from an FE model are to be trusted, they need validating. The extent of this validation often depends on the analyst's knowledge of conducting comparable analyses. A simple type of validation may involve doing hand-calculations for some load cases to check that the output from the FE analysis is as expected. A more involved method of validation involves running a benchmark study, with the response from the FE model compared to experimental results or other trusted data. It is possible (plausible, even) that previously defined parameters such as element type, mesh characteristics, convergence criteria and so on, need modification to pass the validation. Once the model has been sufficiently validated, the load combination(s) of interest may be analysed. The results should be scrutinised using engineering acumen before being thoroughly documented (complemented by plots of the deformed mesh, principal stresses, load-displacement, crack pattern (for NLFEA) and the like).

3.2 Damage caused by corrosion

As stated earlier, corrosion of reinforcement bars is the most common damage mechanism for RC structures. This section therefore gives a brief background on the corrosion process itself and how it affects the structural behaviour. The section then goes on to describe how these effects may be accounted for in structural models.

3.2.1 Corrosion of reinforcement in concrete

Corrosion is an electrochemical process [43] involving oxidation of iron as the anodic reaction and reduction of oxygen as the cathodic reaction. This may be represented by the following half-cell reactions [44]:

$$Fe \to Fe^{2+} + 2e^- \tag{3.1}$$

$$\frac{1}{4}O_2 + \frac{1}{2}H_2O + e^- \to OH^- \tag{3.2}$$

 Fe^{2+} , in turn, reacts with constituents of the pore solution, resulting in other corrosion products. Given the same mass, the corrosion products occupy a volume around two to six times greater than steel [45]. An internal pressure then arises, due to the volumetric expansion as the steel becomes rust. Eventually, the surrounding concrete fails to carry the tensile stresses and splitting cracks develop along the reinforcement, resulting in an increased corrosion rate [46].

In sound concrete, this reaction is prevented by the high alkalinity of the concrete's pore solution and a passivating layer of iron-oxide is formed on the surface of the reinforcement bar [44]. However, this passivation may be broken. Atmospheric carbon dioxide may react with the cement matrix and lower the pH of the pore solutions within the concrete (carbonation), resulting in depassivation of the reinforcement bars. Another common corrosion initiation mechanism is the ingress of chloride ions through the concrete cover (in, say, a marine environment or from de-icing salts). The types of corrosion typically distinguished for reinforcement bars are uniform (or general) and localised (or pitting) corrosion [47]. As the names imply, a uniformly corroded reinforcement bar shows regular loss of material, while localised corrosion occurs at discrete places along the reinforcement bar. Schematically, a high chloride concentration (or carbonation of the concrete) is associated with general corrosion, while lower chloride concentrations are linked to localised corrosion.

Determining the corrosion level of reinforcement in field conditions is a challenging task [48]. Visual inspection may provide information on cracks and delamination but is limited in its ability to evaluate the actual corrosion level. However, based on the results of visual inspection, other methods (such as potential mapping) may be deployed to investigate whether corrosion has started in other parts of the structure but not progressed enough to show visual signs. An estimate of the corrosion level may be obtained non-destructively by measuring the corrosion rate, half-cell potential and resistivity of concrete [49, 50]. However, the associated uncertainties are great due to, among other things, the fact that the start of the propagation period must be known or estimated. Furthermore, research efforts have also been directed towards characterising the corrosion based on measured surface crack widths. See [51] for a comparison of various methods. The model uncertainty is great due to several factors, including environmental exposure and type of corrosion (general or localised) that is forming. The lack of convincing methods to determine the corrosion level of existing structures makes assessment methods which do not need this information attractive. Paper IV investigated the possibility of assessing the structural capacity (concerning anchorage failure in a beam with corrosion-induced cracks along the reinforcement) without information on the corrosion level.

3.2.2 Structural effects of corrosion damage

The structural effects of corrosion for RC members are numerous and concern both the serviceability limit state (SLS) and ultimate limit state (ULS). Figure 3.3 shows an overview of the most important effects [4].



Figure 3.3. An overview of structural effects of reinforcement corrosion, modified from [4].

Corrosion causes a loss of rebar cross-section and a volumetric expansion as steel becomes rust. Furthermore, the corrosion products may create a weak interfacial layer between concrete and reinforcement. Corrosion damage may influence both the ultimate moment and shear capacity, due to several reasons [52]: i) reduction of cross-sectional area of rebar, ii) loss of concrete cross-section and iii) reduction of concrete compressive strength and ductility due to corrosion-induced cracking. Furthermore, the ductility of the reinforcement bars may be impaired due to localised corrosion [53, 54] and the tension stiffening may be affected by the degradation of bond and cracking of the concrete cover. These aspects jointly influence the ultimate deflection, but also the SLS condition in terms of deflection and crack width for service loads. By extension, the plastic rotation capacity may be impaired, which influences the moment redistribution for indeterminate structures, as well as robustness and seismic resistance [4].

The effect of reinforcement corrosion on the bond and anchorage capacity was one of the focal areas of this thesis (addressed in **Paper II** and **IV**) and this structural effect is therefore thoroughly addressed below. Initially, when corrosion of the reinforcement bars propagates, the bond capacity may increase, assuming that the confinement from surrounding concrete is sufficient. With increased corrosion, the tensile hoop stresses in the concrete grow, until they finally crack the concrete cover and longitudinal splitting cracks form. The confinement (and thereby the bond capacity) decreases [6, 55 and 56]. Moreover, the bond may also be decreased by the layer of corrosion products between concrete and rebar. Upon further corrosion, a marked decrease in capacity is expected in case of low levels of transverse reinforcement, while a small increase may be observed in the case of high levels [8, 57–59]. As previously stated, the cross-sectional area is obviously influenced by corrosion and, in some cases, the tensile capacity of the rebar may be the limiting factor on the anchored force. Furthermore, one common method of manufacturing reinforcement bars results in non-uniform strength distribution over the bar's

cross-section, with higher-strength steel close to the surface (as compared to the centre of the bar). Therefore, corrosion may also reduce the average tensile strength of the material [60].

Anchorage failure is typically brittle, whereas a ductile failure mode is preferred since it increases the likelihood of users avoiding imminent danger. Thus, the bond of reinforcement is a particular concern when assessing deteriorated structures and needs to be captured realistically in structural analyses.

3.3 Modelling of corrosion-damaged structures

This section presents important aspects for modelling corrosion-damaged structures. A general overview is given, followed by a description of the bond and anchorage modelling conducted in **Papers II** and **IV**.

Many choices regarding structural modelling relate to the scope of the investigation, cf. the preliminary, detailed assessment in Section 2. Hand calculations with certain simplifying assumptions may be acceptable in some situations; in others, a detailed FE analysis is more appropriate. Various aspects need consideration when comprehensively modelling the influence of corrosion on structural capacity, specifically [52]:

- reduction of cross-sectional area for longitudinal and transversal reinforcement bars;
- reduced ductility of reinforcement bars, due to localised corrosion;
- reduction of cross-sectional area of concrete, due to spalling;
- changes in the constitutive relations (strength and ductility) of concrete, due to cracking caused by expansive corrosion products;
- changed tension stiffening behaviour, due to cover cracking and bond deterioration;
- changed bond behaviour, depending on the corrosion level.

For general corrosion, the cross-sectional reduction of reinforcement bars is simple to implement by changing the geometry. However, quantifying the corrosion level is difficult in practice, as mentioned in Section 3.2.1. Localised corrosion is more complicated to implement (especially since the pit locations and features are typically unknown) and is often simplified in analysis. One of the main influences of pitting corrosion is a reduction in ductility of the reinforcement bars. This effect may be modelled by changing the material properties (cf. [52, 61]). The change in cross-sectional area of the concrete may be considered explicitly in the models. Furthermore, the reduction in compressive strength and ductility of the concrete (due to cracks parallel to the principal compressive direction) may be considered by a modified stress-strain relationship. For example, the relationship proposed by Vecchio and Collins [62], is readily available for use in analyses, at different levels of detail. In more detailed analyses, the changed tension stiffening behaviour may be represented by adjusting the bond stress-slip relationship for the reinforcement-to-concrete interaction. If the ULS is a concern when making the calculations, the effect of tension stiffening is not influential and may be omitted from, say, hand calculations.

The modelling of bond and anchorage has been addressed by several researchers. See [63–65] for examples of simplified models and [11, 12 and 66] for more advanced models. In contrast to the proposal in **Paper II**, these simplified models do not include a bond stress-slip relationship. Rather, numerical expressions (calibrated to experimental data) are used to obtain the reduction in capacity as a function of the corrosion level. The advanced models are implemented in 3D FE modelling and are capable of producing more realistic results for the

bond behaviour. However, they require major effort to implement. Furthermore, input parameters (such as the corrosion level) or the parameters needed to simulate it (relating to such things as material properties, geometry and exposure), are required for all these models. In this thesis, modelling bond behaviour and anchorage capacity were the topics of **Papers II** and **IV**, with **Paper II** presenting a simplified model and **Paper IV** an advanced one. The benefit of using these models is that the full bond stress-slip relationship is obtained in **Paper II** (for use in, say, FE analyses) and, in **Paper IV**, it was possible to estimate the anchorage capacity without knowing the corrosion level.

The simplified model in **Paper II** assesses the anchorage capacity by solving the equilibrium conditions along the reinforcement bar, as described by the differential equation:

$$\frac{\pi \cdot \phi_{\rm m}^2}{4} \cdot \frac{d\sigma_{\rm s}}{dx} - \pi \cdot \phi_{\rm m} \cdot \tau_{\rm b} = 0 \tag{3.3}$$

where $\phi_{\rm m}$ is the reinforcement diameter, $\sigma_{\rm s}$ is the stress in the reinforcement, $\tau_{\rm b}$ is the local bond stress and x denotes the longitudinal direction of the bar. The local bond stress-slip relationship constitutes the core of the model and was based on the relationship in *fib* Model Code 2010 [16] but with some modification and additions, namely:

- introduction of equivalent slip to account for bond degradation due to corrosion;
- change of failure mode due to corrosion-induced cracking of the concrete cover;
- modification of the residual bond stress in case of low stirrup content.

The features bulleted above are illustrated in Figure 3.4. The equivalent slip, s_{eq} , was calibrated against a large database of bond tests and the residual bond stress for low stirrup content was modified to reflect the experimental data for beams found in the literature. For an exhaustive explanation of the model, see **Paper II**.



Figure 3.4. a) Concept of equivalent slip illustrated on bond stress-slip relationship, b) change of failure mode due to corrosion-induced cracking. For a corrosion level W_c above the cracking limit W_{ct} , the splitting strength was reduced. Note also that the residual bond stress is greater than zero for cases without stirrups. Adopted from **Paper II**.

Paper IV investigated various methods of assessing the capacity of reinforcement bars anchored in concrete with corrosion-induced cracks. The most promising, advanced method is presented here. It is based on an NLFE analysis, in which both the concrete and reinforcement

are 3D-modelled. A friction model was used for the interface between concrete and reinforcement, to explicitly model the confinement from the surrounding concrete. The effect of corrosion-induced cracks was represented by locally weakening the concrete's tensile properties for the FEs at the crack position, thereby reducing the confinement of the reinforcement bar. The tensile properties were weakened based on the measured crack width, using the bi-linear softening relationship assumed for the concrete in tension. For further information, see Section 3.5. A graphical representation of the weakened FEs for a beam is shown in Figure 3.5.



Figure 3.5. Elements assigned weakened elements on a) the left-hand and b) the right-hand side of a beam specimen. Adopted from **Paper IV**. Note that the reinforcement bars were debonded, except for the outer 100 mm on each side, similar to the experiment being modelled.

3.4 Cracking in reinforced concrete

As mentioned in the introduction, cracking in RC structures is common and not necessarily detrimental. Many structures are designed to crack under service loads; this is necessary if mild reinforcement is to carry any significant tensile force. Current structural design codes specify crack width limitations and minimum reinforcement levels, to control cracking (cf. [67]) and meet durability and aesthetic requirements. In practice, cracks may arise for numerous reasons at many stages of a structure's life. **Papers IV** and **V** investigate how cracks formed due to corrosion and external loading (respectively) may be included in structural analyses. An overview in this section provides background and context for cracking in concrete, as illustrated in Figure 3.6.



Figure 3.6. Overview of different types of cracks in concrete, adopted from [68].

The different types of cracks may be sorted into those occurring before and after hardening. Before hardening, cracks may occur due to plastic shrinkage or settlement, or due to movement of the subgrade. Early frost damage may also induce cracks before the concrete has hardened. After hardening of the concrete, several causes of cracking may be identified, including physical, chemical, thermal and structural. Physical cracks may be observed as crazing on the concrete surface, or drying shrinkage when contraction of the concrete is prevented by internal or external restraints. Moreover, certain types of aggregates may shrink and give rise to cracking. Cracks may also form due to chemical reactions such as corrosion of the reinforcement (addressed in Section 3.2), alkali-aggregate reactions or carbonation of the concrete. Thermal causes of cracking also include freeze-thaw cycles and external seasonal temperature variations as well as early-age thermal contraction (externally or internally restrained). Lastly, cracking due to structural causes may arise from loading to design levels, accidental overloading and stress-dependent strain due to creep.

Cracks may be sorted into two main categories; structural cracks and non-structural cracks [68]. The label "structural" is used to describe cracks developed due to the load-carrying mechanisms of the concrete structure. For example, external loading of a member will cause a certain stress distribution within the structure. When these stresses exceed the tensile strength of concrete, structural cracks will form. This category of cracks is also the most likely to influence structural behaviour. Non-structural cracks, on the other hand, are not directly related to the load-carrying mechanisms of the structure and include cracks formed due to restrained shrinkage and thermal movement. Most non-structural cracks are unlikely to influence the short-term structural behaviour, although they may do in the long term (in the case of, say, internal frost attacks).

However, it should be noted that cracks due to reinforcement corrosion might affect the loadcarrying capacity if anchorage of the bars is a limiting factor. Moreover, determining the exact causes of cracking very often requires thorough investigation. With knowledge of the loadcarrying system, it is sometimes possible to distinguish between different types of cracks based on their location, appearance and the time of their formation. This information may be complemented with concrete core samples from the structure. These may be used to prepare "thin sections" which provide further information on the mechanism of cracking, visually or via microscopy [69].

In the experiments analysed in **Papers IV** and **V**, different methods were applied to induce cracks in the test specimens. In **Paper IV**, reinforcement bar segments in both anchorage regions of a simply supported beam were artificially corroded, to induce splitting cracks along the reinforcement. By contrast, the cracks in **Paper V** originated from restraint shrinkage with subsequent tensile loading: this resulted in crack planes with normal directions coinciding with the length direction of the beam. The next section presents methods for including cracks such as these in structural models.

3.5 Incorporating pre-existing cracks in FE modelling

At the time of assessment, the structure may be cracked due to one or more of the mechanisms presented in Section 3.4. If structural behaviour is to be realistically modelled, it may be important to include cracks in the analysis. Within this work, the term "pre-existing cracks" was chosen to describe cracks present in the structure at the start of the FE analysis, in contrast to the simulated cracks developing during the analysis. It is worth noting that, in practice, including all cracks in the analysis will often be unfeasible. Based on an understanding of the structural system and the location and characteristics of the cracks, engineering judgement should be used when identifying critical cracks for inclusion in the analysis.

In a standard NLFE analysis of reinforced concrete, the two most common methods of modelling cracks are the smeared and discrete concepts [70]. The smeared crack approach treats a cracked solid as a continuum, while the discrete-crack approach treats a crack as a geometrical discontinuity. While smeared crack models often employ a total-strain-based approach with either fixed or rotating cracks, properties of the discrete cracks are defined based on their relative displacements in the normal and transverse directions. Discrete cracks are commonly used to incorporate pre-existing cracks in assessments of dams [69], where they are assigned frictional properties. In such large structures, it is feasible to omit the aggregate interlock in the shear transfer and only consider frictional mechanisms. However, this is not generally acceptable in other structures, such as concrete beams with low levels of shear reinforcement [71]. Another example of how discrete cracks are used may be found in [72]; in this instance, they were used to model through-cracks in RC slabs tested by bending, for fatigue assessment. In this case, the treatment of aggregate interlock was tailored to represent the fatigue response of the slab and was not generally applicable.

Moreover, examples in which the smeared crack approach has been used to incorporate preexisting cracks are scarce in the literature, with only a few works found. In an assessment of concrete columns with pre-existing cracks [73], a tailored load arrangement was applied in analyses to mimic the pre-existing cracks before the loading of interest was subsequently applied. Although this approach represented the pre-existing cracks and the maximum loading reasonably well, the implementation was deemed cumbersome and the approach

CHALMERS, Architecture and Civil Engineering

computationally expensive for the treatment of multiple cracks. Furthermore, a methodology for implementing cracks also appears in [74]. There, the constitutive relationship for the cracks was derived from the uncracked state, with a scalar damage parameter to reduce the stiffness and strength. The influence of different crack widths was discussed. However, this was not implemented in the analyses, where the same damage parameters were used for all cracks. The proposed methodology in [74] has similarities to the models presented in **Papers IV** and **V**, in that the constitutive relations of the continuum elements are weakened to resemble pre-existing cracks. However, the derivation of damage parameters is different. The work included in this thesis develops a modelling methodology to incorporate pre-existing cracks in FE analysis and addresses weak points identified in the literature, by making the damage dependent on the individual crack width and including aggregate interlock as shear retention.

Papers IV and V investigated two advanced approaches for including pre-existing cracks in FE analysis. These were: a) weakening the continuum elements at the position of the crack, and b) introducing discrete crack elements with weakened properties. Approach a) was deemed the most promising in both papers and is presented here. The reader is referred to **Papers IV** and **V** for a presentation of approach b).

In the weakened elements approach, the FEs at the positions of the pre-existing cracks are assigned weakened tensile properties, as compared to the intact concrete. A bilinear, mode-I, stress-to-crack width relationship for undamaged concrete served as a basis for deriving the weakened tensile properties. Based on a measured crack width, the weakened properties (in terms of tensile strength) and remaining crack opening until stress-free crack, were derived for the specific crack from the bilinear relationship. The procedure is shown in Figure 3.7. By using the measured crack width (w_c) in the bilinear tensile stress-to-crack opening relationship for undamaged concrete, the tensile stress ($f_{ct,c}$) and residual fracture energy (G_{F,w_c}) were determined. Furthermore, the corresponding stress-strain relationship, shown in the middle of Figure 3.7, was derived using the modulus of elasticity (E_{cm}) and an assumed crack bandwidth (h). In determining the crack bandwidth, the cracks were assumed to be localised within one weakened element row. The ultimate strain ($\varepsilon_{w_{c,ult}}$) in the stress-strain relationship (used as

input for the weakened elements) was determined so that the area under the stress-strain relationship equalled the fracture energy divided by the crack bandwidth $(G_{F,w_c}/h)$, meaning:

$$\varepsilon_{w_{\rm c,ult}} = \frac{w_{\rm ult} - w_{\rm c}}{h} \text{ for } w_{\rm c} > w_{\rm s}$$
(3.4)

Moreover, to prevent interaction of the strains in different directions, a Poisson's ratio of zero was used for the weakened elements. In some cases, the measured cracks widths were also larger than w_{ult} . In such cases, low tensile properties were assigned by using a crack width of $0.99w_{ult}$ in the calculations.



Figure 3.7. Principal idea for weakening tensile properties of FEs representing cracks, from Paper V.

The fixed-crack approach, in which a "shear-retention factor" specifies the remainder of the initial shear modulus, was used for the weakened elements. The shear-retention is influenced by several factors, such as the type of aggregates used. The choice of suitable shear-retention for analysis is not trivial. Guidelines for NLFE analysis [20] stipulate variable ("damage-based") shear retention which decreases after cracking, based on the increase in normal strain. This type of shear retention was used for the sound concrete material in **Paper V**. However, a fixed shear-retention factor of 0.01 was used for the weakened elements, to compare the results from analyses of weakened elements and discrete cracks. This is because the analyses were conducted in the commercial FE software DIANA 10.3 [42], in which variable shear retention for discrete cracks is not available. The choice of shear retention is also thoroughly discussed in **Paper V**.

An example of how this method may appear in the FE model is shown in Figure 3.8. This shows a surface crack pattern for side 1 of specimen PC1 in **Paper V** in a), with blue indicating cracks formed in the restrained shrinkage phase and red indicating cracks formed during tensile loading. The widths of the cracks may be found in **Paper V** using the notations indicated in the figure. The cracks on each surface were extruded towards the centerline of the beam, with oneelement imbrication at the mid-section. The FEs coinciding with the extruded cracks were assigned weakened tensile properties using an automated selection procedure. Furthermore, additional elements were manually weakened to allow the path of the weakened elements to open without cracking sound concrete elements. The FE mesh with weakened (cracked) elements in red for side 1 of PC1 is shown in b) and the top side is shown in c). The manually selected elements in b) and overlapping elements in c) are indicated in orange.



Figure 3.8. Example of incorporating pre-existing cracks in an FE mesh, modified from **Paper** V. a) shows the surface crack pattern for side 1 of specimen PC1, b) shows side 1 of the FE mesh with weakened elements in red and orange and c) shows the top of specimen PC1. The manually selected elements in b) and the overlapping elements in c) are indicated in orange.

It should be noted that some types of cracks most likely necessitate further modifications in the analysis. For example, representing frost damage requires modification of the compressive strength [75] and (in the case of corrosion) the reduction of reinforcement cross-sections must be considered for the bending and shear failure modes. This thesis has studied corrosion-induced cracks in the anchorage region and cracks from restrained shrinkage and tensile loading. It was therefore deemed reasonable to modify only the tensile properties of the elements representing the cracks.

4 Background on structural reliability assessment

Within this thesis, **Paper I** evaluated safety formats and **Paper III** derived partial factors. To set this work in the context of structural reliability assessment, this section provides a brief background of the subject and explains some important concepts. Specific presentations will be given of limit state design, uncertainties in basic variables, reliability level and deterministic and probabilistic means of ensuring the target reliability level is reached. These procedures are related to the verification stage shown in Figure 2.1.

4.1 Principles of limit state design

Internal stresses will arise in a structure when subjected to applied loading. The response, in terms of stress distribution and magnitude, depends on the characteristics of the load as well as the stiffness, strength and shape of the structure. "Limit states" are defined, to determine whether the response from the structure is acceptable. The two main limit states are presented more thoroughly below; namely, the ultimate limit state (ULS) and serviceability limit state (SLS).

ULS concerns the maximum load-carrying capacity of a structure but also includes the maximum deformation capacity. Its nature is typically irreversible, resulting in failure at the first violation. According to Eurocode 1990 [15] and the Joint Committee of Structural Safety (JCSS) probabilistic model code [76], the ULS may be exemplified by the following situations:

- excessive deformation, transformation of the structure or any part of it into a mechanism, or rupture;
- loss of equilibrium in the entire structure or part thereof, causing rigid body movement;
- sudden change of structural system, such as snap-through behaviour;
- time-dependent failure, such as rupture of connections or members due to fatigue.

SLS reflects the service requirements put on a structure. According to JCSS [76], a few examples are:

- local damage, possibly affecting the durability or appearance of an element in the structure. For example, overly wide cracks in concrete;
- excessive deflections;
- observable damage due to time-dependent effects, such as fatigue;
- disallowed deformations, unacceptably impairing the functionality or appearance;
- excessive vibrations, affecting people's comfort and the function of equipment.

In contrast to ULS, an SLS violation is quite often reversible. For example, excessive deflections caused by external loading may decrease to allowable levels once the loading is reduced. Concerning SLS, the specific failure limit values should be stipulated based on utility considerations [76].

Assessment of structural reliability aims to find the probability of limit-state violation at any point in a structure's life [77]. The more specific term *structural safety* assessments concerns ULS in particular (a concern addressed by the work of **Papers I** and **III**). In practice, it is common to use a combination of frequency observations and subjective estimates of structural elements and properties to determine the probability of a limit state being violated.

CHALMERS, Architecture and Civil Engineering
4.2 Uncertainty of basic variables

Basic variables describe the performance and safety of a structure and include physical properties related to such things as materials, loads and dimensions. [77]. A broad definition also includes parameters characterising the model itself, with the basic variables assumed to contain all the necessary information for input into the calculation model [76].

A basic variable may be described as a random variable with a suitable probability distribution. A particular distribution is often chosen based on observed data and experience, or taken as normally distributed based on central limit theorem [77]. Random processes (random variables evolving) and random fields (spatial differences) may also be used to describe basic variables. Examples of standardised probabilistic representations of common basic variables (such as the material properties of concrete) may be found in [76].

The representation of basic variables must include uncertainties from all key sources. Three main types of uncertainty may be distinguished [76]:

- inherent physical or mechanical uncertainty;
- statistical uncertainty;
- model uncertainty.

Inherent physical or mechanical uncertainty may be represented by the variation in material strength and variation in geometrical properties. This uncertainty is considered by the probability distribution representing the basic variable. Moreover, statistical uncertainty arises when using statistical estimators (such as sample mean and standard deviation) to determine a suitable probability density function and related parameters. This is because the observations rarely represent the variable perfectly and because different statistical estimators are typically obtained based on the choice of sample data set [77]. Furthermore, the modelling uncertainty reflects the fact that the results obtained from a model will be more or less incomplete and inexact (due to such things as simplifications or lack of knowledge). Note that several types of uncertainties may be (and typically are) present for the same basic variable. For example, consider load models, in which the probability density functions describing the load magnitudes are based on statistical data and therefore marked with statistical uncertainty. However, these are also simplifications of reality and carry modelling uncertainties. To illustrate the meaning of modelling uncertainty, consider the following calculation model for general random variables [76]:

$$Y = f(X_1, X_2, \dots X_n)$$
(4.1)

where Y is the model output, f() is the model function and $X_1, X_2, ..., X_n$ are the basic variables. The model function, f(), is typically not complete and exact. Therefore, the output Y cannot be perfectly predicted, given a set of basic variables X_i in an experimental setting. The model may instead be expressed:

$$Y = f(X_1 \dots X_n, \theta_1 \dots \theta_m)$$
(4.2)

where $\theta_1 \dots \theta_m$ are random variables containing information about the model's uncertainty. Their statistical properties are typically determined from observations or experiments and the mean values should be set so that, on average, Y is predicted correctly. However, it should be noted that the model's uncertainty may be difficult to assess in some cases. For example, when the correct failure mode is difficult to capture in the numerical analysis, as shown by [78].

4.3 Target reliability level

When conducting a safety assessment (**Paper I**) or deriving partial safety factors (**Paper III**), the intended reliability of the structure needs to be expressed. A widely used measure of reliability is the generalised reliability index β [76]. This is based on the standard normal (or Gaussian) distribution, in which the mean value is zero and the standard deviation is one. The expression for β may be written:

$$\beta = -\Phi^{-1}(p_{\rm f}) \tag{4.3}$$

where p_f is the probability of failure and Φ^{-1} is the inverse of the normal cumulative distribution function. Consider the limit-state function Z = R - S, where resistance R and load effect S are normally distributed. β may then be understood as the number of standard deviations away from the mean value, resulting in the area p_f under the curve in the unsafe domain ($Z \le 0$), as shown in Figure 4.1.



Figure 4.1. Illustration of β in relation to p_f for Z = R - S, where resistance R and load effect S are normally distributed (modified from [77]).

A structure is designed to meet the ULS and SLS requirements. However, since uncertainties are present in the basic variables and calculations models, the probability of failure cannot be zero. Rather, the design should comply with a target reliability index (directly equivalent to a probability of failure). In one sense, this may be viewed as an optimisation problem where the optimal β may be found, based on the failure cost (related to the consequences of failure) and the cost of increased safety. However, such an approach may be ethically questionable in cases when aspects relating to fatalities, injuries and culture take precedence over economic loss. Therefore, risk-benefit analyses are often used to relate the reliability of the structure to the cost per life saved. This cost may be chosen by comparison with other, similar structures; it forms the basis for determining the target reliability. There are various recommendations for target reliabilities, such as the ones by Eurocode [15], the International Organisation for Standardization (ISO) [34] and the Joint Committee on Structural Safety (JCCS). The latter's recommendations for target reliability in ULS [76] are shown in Table 4.1. Allegedly, their basis was public-facing cost-benefit analyses of simple but representable sample structures.

Relative cost of safety measure	Consequences of failure		
	Minor	Moderate	Large
Large (A)	3.1 (10 ⁻³)	3.3 (5.10-4)	3.7 (10-4)
Normal (B)	3.7 (10-4)	4.2 (10 ⁻⁵)	4.4 (5.10-6)
Small (C)	4.2 (10 ⁻⁵)	4.4 (5.10-6)	4.7 (10 ⁻⁶)

Table 4.1. Tentative target reliability indices β (probability of failure in parentheses) related to the ULS for a one-year reference period, adopted from [76].

For a complete explanation of the table, the reader is referred to the probabilistic model code [76]. In simple terms, the consequences of failure include the risk of casualties in case of failure as well as the economic consequences, while the "relative cost of safety measure" considers the cost of increasing the safety. For **Paper I**, a target reliability of 3.8 for a 50-year reference period was used, as stipulated in Eurocode [15] for the design of new, class RC2 structures. However, a target reliability of 3.7 for a one-year reference period was used in **Paper III**. This corresponds to a case which had large consequences of failure and a large relative cost of safety measure in Table 4.1. This was considered reasonable for an existing structure at the end of its service life.

4.4 Measures of structural reliability

Measures of structural reliability are used to safeguard the target reliability within a structural design or assessment. The overview presented here is based on [77]. In structural engineering practice, a simplified measure of limit-state violation is typically deployed. The reliability is checked by using a deterministic representation of the basic variables, derived from their probabilistic distributions, directly in the calculations. These deterministic methods have developed over time and the most common current procedure is to use partial safety factors. This measure of safety was developed by combining two others; the *factor of safety* and the *load factor* measure. The factor of safety approach involves reducing the material capacity on the sectional level, while the load factor approach entails upscaling the loads on the structural level. The resulting equation for limit state i may be written:

$$\frac{R}{\gamma_{R,i}} \le \gamma_{D,i} \cdot S_{D,i} + \gamma_{L,i} \cdot S_{L,i} + \cdots$$
(4.4)

where *R* represents the member resistance and γ_R the corresponding partial safety factor. The load effects from dead and live loads are represented by S_D and S_L , respectively and the corresponding partial safety factors are γ_D and γ_S . The partial safety factor values are calibrated to result in a design with the intended safety level. The simple nature of this safety measure, (while still allowing the different uncertainties related to different basic variables to be represented) has made it widely used within structural engineering. In **Paper III**, partial safety factors were derived for the model that was used to assess the anchorage of corroded reinforcement (**Paper II**).

The partial safety factors developed in **Paper III** were verified, to confirm that they led to a design with the intended probability of failure. This was done by using a probabilistic measure of limit-state violation, to determine the probability of failure for several designs made using the partial safety factors. Again, as a simple example, consider a structure with resistance R, subjected to a load effect S (both independent random variables). The probability of failure may then be expressed:

$$p_{\rm f} = P(R \le S) = P(R - S \le 0) = P[G(R, S) \le 0]$$
(4.5)

where p_f is the probability of failure, P() simply means the probability and G() is a general expression of the "limit-state function". To calculate the probability of failure, the bivariate (joint) probability density function (*pdf*) of *R* and *S* may be used:

$$p_{\rm f} = P(R - S \le 0) = \iint_D f_{RS}(r, s) \, dr ds$$
 (4.6)

where D denotes the failure domain. Since R and S were assumed to be independent, this may be written:

$$p_{\rm f} = \int_{-\infty}^{\infty} \int_{-\infty}^{S \ge r} f_R(r) f_S(s) \, dr ds \tag{4.7}$$

remembering that the cumulative distribution function (*cdf*) is defined as:

$$F_X(x) = P(X \le x) = \int_{-\infty}^x f_X(y) \, dy$$
 (4.8)

Combining Eqs. (4.7-4.8), the probability of failure may be written as a single integral:

$$p_{\rm f} = \int_{-\infty}^{\infty} F_R(x) f_S(x) \, dx \tag{4.9}$$

Eq. (4.9) basically expresses the probability of the resistance being lower than the load effect and integrates over the entire domain to obtain the probability of failure. A visual representation of the load and resistance, and the probability of failure, is shown in Figure 4.2.



Figure 4.2. Probability density functions of load and resistance, with visual interpretation of probability of failure. Adopted from [77].

Although a simple case is suitable for instructional reasons, to be useful in practical applications it must be generalised. This because, in many cases, both the resistance and load effect are functions of several random variables. They may also be dependent, meaning that the load influences the resistance (as with a beam subjected to bending and normal forces, for example). If the limit-state function is described by G(X), the probability of failure may be written:

$$p_{\rm f} = P[G(X) \le 0] = \int \dots \int_{G(X) \le 0} f_X(x) \, dx \tag{4.10}$$

where $f_X(x)$ is the joint *pdf* of the n-dimensional vector X containing all the basic variables. Note that the terms R and S are not explicitly included, but rather are implicit in X. In general, the probability of failure in Eq. (4.10) cannot be obtained by integrating analytically over the failure domain $G(X) \leq 0$. The next section presents approximate methods for evaluating the probability of failure.

4.5 Structural reliability methods

To evaluate the probability of failure, Eq. (4.10) must be solved. However, no analytical solution is available in practice since both integrand and failure domain may be quite complex. Two main approaches have been developed to overcome this challenge; approximation methods and simulation methods [79]. In approximation methods, a local approximation of the limit-state function is found at a point of interest. Simulation (or Monte Carlo) methods, on the other hand, numerically estimate the multidimensional integral based on input sampled from the basic variables. In **Paper I**, the probability of failure was estimated by approximation methods, with the first-order reliability method (FORM) applied to response surfaces established by

evaluating the limit-state function. **Paper III** used a combination of approximation and simulation methods; a FORM analysis was conducted, with subsequent "importance-sampling" simulations in *UQLab: A framework for Uncertainty Quantification in MATLAB* [80].

A brief introduction to the methods used in **Papers I** and **III** is given below. For a detailed overview of common structural reliability methods, the reader is referred to [77, 81 and 82], among others.

4.5.1 First-order reliability method

A common approximation approach is the first-order reliability method (FORM). This involves three main steps, to approximate the probability of failure [79]:

- transformation of the input vector of basic variables $X \sim f_X(x)$ into a standard normal vector $U \sim N(\mathbf{0}, I_M)$, where I_M contains the ones pertaining to the standard deviation in an *M*-dimensional case;
- a search algorithm is used to find the design point U^* , corresponding to the point in standard normal space where failure is most likely to occur;
- linear approximation of the limit-state function at U^* and subsequent calculation of the resulting estimated probability of failure p_{f} .

In the first step, all the variables are transformed to standard normal distributions (that is, with zero mean and unit standard deviation). If the basic variables are uncorrelated and normally distributed, the transformation is simply [77]:

$$U_{i} = \frac{x_{i} - \mu_{X_{i}}}{\sigma_{X_{i}}} \tag{4.11}$$

where μ_{X_i} and σ_{X_i} are the mean value and standard deviation of the basic variable X_i , respectively. If the basic variables are correlated, or if they follow distributions other than the normal ones, the transformation becomes more involved and is not addressed here. Rather, the reader is referred to [77]. With the basic variables transformed to the standard normal space, the joint probability density function $f_{U}(u)$ is the standard multivariate normal distribution. Logically, the limit-state function should also be transformed, thus leading to the failure definition $g(u) \leq 0$, see Figure 4.3.

The point on the failure (hyper)surface closest to the origin in standard normal space is referred to as the design point (or checking point) U^* . Since the basic variables and limit-state function were transformed to standard normal space, this is also the most probable point of failure. In mathematical terms, the design point may be defined [79]:

$$\boldsymbol{U}^* = \underset{\boldsymbol{u} \in \mathbb{R}^M}{\operatorname{argmin}} (\|\boldsymbol{u}\|, g(\boldsymbol{u}) \le 0)$$
(4.12)

In many practical cases, the limit-state function is non-linear and U^* needs to be solved in an iterative procedure. There are various algorithms to this end, such as the Hasofer-Lind-Rackwitz-Fiessler algorithm and improvements thereto [83–85]; the reader is referred there and to [77] for an explanation of the iterative procedure. Note that the limit-state function is approximated by a first-order Taylor series expansion in the iterative procedure to find U^* ; it is therefore called the first-order reliability method.



Figure 4.3. a) Basic variables and limit-state function in physical space, b) basic variables and limit-state function transformed to standard normal space with the design point \mathbf{U}^* and the Hasofer-Lind safety index β_{HL} indicated (modified from [79]).

The Hasofer-Lind safety index β_{HL} [83] may be calculated based on U^* and α^* , where the asterisk indicates the converged design point, as follows:

$$\boldsymbol{\beta}_{\mathrm{HL}} = \boldsymbol{\alpha}^* \cdot \boldsymbol{U}^* \tag{4.13}$$

This may be used to find the probability of failure through Eq. (4.3). See Figure 4.3 b) for a graphical representation of β_{HL} and U^* . The direction cosines (collected in α^*) represent the sensitivity of the limit-state function to changes of the basic variables [77]. The direction cosines squared (α_i^2 for the ith basic variable) are called "sensitivity factors" and total to unity. These may have important practical application since basic variables associated with high sensitivity levels need to be accurately described. Those associated with low sensitivity may not need the same level of accuracy and, to reduce the dimensionality of the problem, may even sometimes be deemed deterministic.

4.5.2 Monte Carlo simulations

The other type of main approach to estimating Eq. (4.10) is called "Monte Carlo simulations" or just "simulations". It involves random (or rather pseudo-random) sampling of values from the probabilistic distributions of the basic variables, so as to conduct numerous evaluations of the limit-state function. The probability of failure may then be calculated as:

$$p_{\rm f} \approx \frac{n(G(x_{\rm i}) \le 0)}{N} \tag{4.14}$$

where $n(G(\mathbf{x}_i) \leq 0)$ denotes the number of limit-state violations and N is the total number of evaluations. Logically, N governs the accuracy of the estimate of p_f and an increase in the number of evaluations increases the accuracy.

Sampling basic variables

The most straightforward way to sample from the basic variables is to generate "random variates" from their probabilistic distributions. This may be achieved by such means as the

inverse transform method, in which a random value between zero and one is generated and the inverse of the cdf is used to obtain the random variate, see Figure 4.4.



Figure 4.4. Generation of random variates by the inverse transform method, modified from [77].

This type of sampling is referred to as "crude Monte Carlo" sampling. However, since the probability of failure is typically very small (cf. Table 4.1), evaluating the limit-state function using this sampling procedure seldom results in failure. Therefore, random variates of the basic variables need to be generated many times (with subsequent evaluation of the limit-state function) to obtain a good representation of the probability of failure. This makes the procedure very expensive in terms of computations and time [86]. As a rule of thumb, about 10,000-20,000 simulations are required for a confidence interval of approximately 95%, depending on the function being evaluated [87].

Variance reduction by importance sampling

As described, crude Monte Carlo requires many simulations to obtain an accurate estimate of $p_{\rm f}$ (that is, with low variance). This is a substantial drawback and hinders practical application [77]. However, techniques for reducing the variance of the probability of failure estimate have been developed; these are known as "variance reduction" techniques [88]. Given the same desired accuracy of $p_{\rm f}$, such techniques may reduce N.

Information about the problem at hand may be used to reduce the variance [77]. Consider the probability density functions (representing R and S) in Figure 4.2. In crude Monte Carlo sampling, random variates are based on their probability density, which makes values in the centre more likely than values in the tail regions. This makes limit-state violations unlikely. However, if sampling is conducted from the overlapping region between R and S, the likelihood of interesting results increases. This is the basic principle behind importance sampling; the samples are generated from probabilistic distributions other than those of the basic variables. This may be described by:

$$p_{\rm f} = \int \dots \int I[G(\mathbf{x}) \le 0] \frac{f_x(\mathbf{x})}{h_v(\mathbf{x})} h_v(\mathbf{x}) \, d\mathbf{x} \tag{4.15}$$

where $I[G(\mathbf{x}) \leq 0]$ indicates limit-state violation, $h_{\nu}(\mathbf{x})$ is the "importance sampling" density function. Choosing a suitable sampling function may be difficult in some cases. However, an efficient function may be found, based on results from a previous FORM analysis [77]. For further treatment of the sampling function, please see [77, 79].

4.5.3 Response surfaces

When structural reliability methods are to be used for implicit limit-state functions, such as FE analyses, the response surface method may be useful. A response surface is constructed based on multiple evaluations of the implicit limit-state function [77]. In practice, this is done by pointwise exploration of the safe and unsafe domains of the implicit limit-state function, using deterministic input values for the basic variables. Thereafter, a polynomial or other type of function is fitted to the set of results.

A tractable simplification of the implicit limit-state function is thus created and may be used for both approximation and simulation approaches. The accuracy of the response surface should be highest in the proximity of the design point, as it is the most probable failure point; a lower level of accuracy may be accepted elsewhere. For a thorough explanation of the procedure please refer to [77, 89 and 90].

5 Summary of appended papers

This section summarises the appended **Papers I** to V to provide their most central content in condensed form. The focus is on the key features of the methodology and the most important results.

Paper I. Evaluation of safety formats for non-linear finite element analyses of statically indeterminate concrete structures subjected to different load paths.

In **Paper I**, the safety formats available in *fib* Model Code 2010 [16] for non-linear finite element analyses (NLFEA) were evaluated for a frame structure subjected to two different load histories. The loading consisted of a vertical point load in the middle of the span and a horizontal point load on the top left-hand corner. The two load histories were defined as application of the vertical load followed by the horizontal load and vice versa, see Figure 5.1.



Figure 5.1. a) frame and load application, b) load histories, in which the main load history follows 1, 3, 4 and the inverse follows 2, 3, 4.

The resistance was defined as the total load at failure, while the safety assessment was based on the limit-state function:

$$g(\mathbf{X}) = R(\mathbf{X}) - R_{\rm SF} \tag{5.1}$$

where $R(\mathbf{X})$ was the resistance function, obtained by fitting response surfaces to multiple NLFEA results and R_{SF} was the design resistance, according to the studied safety format. In the present study, the basic variable \mathbf{X} included the concrete compressive strength and reinforcement yield strength. The first-order reliability method (FORM) was used to evaluate the probability of failure of the limit-state functions constructed for each load history and safety format. A 3D representation of the limit-state function in case of the main load history and the ECOV safety format is shown in Figure 5.3, plus a contour plot of the limit-state function and the FORM design point.



Figure 5.2. Deformed shape and crack pattern for the frame, loaded with both loads to their characteristic level, adopted from **Paper I**. The figure shows a) the main load history, and b) the inverse load history.



Figure 5.3. a) 3D plot of the limit-state function for the main load history and ECOV safety format, b) contour plot of the stated limit-state function, with the FORM design point indicated by a diamond. The two solid lines indicate the mean values of concrete compressive strength and yield strength of reinforcement. Adopted from **Paper I**.

The main findings from **Paper I** may be summarised as follows:

- the load history was shown to be influential for the load-carrying capacity as well as the crack pattern. The latter may be observed in Figure 5.2;
- response surfaces and FORM were applied to study the reliability of the three safety formats (ECOV, GRF and PSF) for the two load histories, see Figure 5.3;
- the ECOV safety format did not reach the intended safety level for the main load history;
- a more detailed definition of the structural resistance was requested, for use with the safety formats.

Paper II. Engineering bond model for corroded reinforcement.

In **Paper II**, a bond model (named ARC2010) for assessing the anchorage capacity of corroded reinforcement bars was further developed and validated against a large test database. The bond model determined the anchorage capacity by solving the equilibrium conditions along the reinforcement bar, as described by the differential equation:

$$\frac{\pi \cdot \phi_{\rm m}^2}{4} \cdot \frac{d\sigma_{\rm s}}{dx} - \pi \cdot \phi_{\rm m} \cdot \tau_{\rm b} = 0 \tag{5.2}$$

where $\phi_{\rm m}$ is the reinforcement diameter, $\sigma_{\rm s}$ is the stress in the reinforcement, $\tau_{\rm b}$ is the local bond stress and x denotes the longitudinal direction of the bar. The local bond stress (as a function of slip between reinforcement and concrete) constituted the core of the model. This was based on the relationship in *fib* Model Code 2010 [16] but some elements were added and modified, specifically:

- introduction of equivalent slip, to account for bond degradation due to corrosion;
- change of failure mode due to corrosion-induced cracking of the concrete cover;
- modification of the residual bond stress, in case of low stirrup content.

See Figure 3.4. for an illustration of these bulleted features.

The database used to calibrate the bond model has been depicted in terms of its size and distribution, as well as the variation of relative bond strength among subsets of test results, see Figure 5.4. In Figure 5.5, the relative average bond strength obtained from the ARC2010 model (in other words, the deteriorated bond strength normalised by the non-corroded bond strength) is plotted against the corresponding database values for cases without and with stirrups. For ARC2010, increases in relative bond strength were deliberately not allowed. However, a number of tests in the database show increased capacity. The scatter among the bond tests was visibly quite large. However, it may also be noted that the results are distributed equally along the diagonal line. This represents the same result in the database and ARC2010 but with a slight skew towards the safe side.



Figure 5.4. a) Four bond index (confinement) intervals and five corrosion level intervals on the horizontal axes and number of test results in each group on the vertical axes and b) contour plot of the coefficient of variation for relative bond strength in the database for varying bond index and corrosion level. Adopted from **Paper II**.



Figure 5.5. Relative average bond strength (from database) plotted against ARC2010 values for cases a) without stirrups and b) with stirrups. Adopted from **Paper II**.

The main findings from Paper II may be summarised as follows:

- a model for assessing anchorage of corroded reinforcement was proposed. This included the confinement effects from concrete and transverse reinforcement, plus the changed confinement at the point where corrosion cracks the concrete cover;
- the model was shown to represent the physical behaviour well, with a marked decrease in bond strength at the point where corrosion cracks the concrete cover (in cases of low stirrup contents). For higher stirrup contents, the decrease was less pronounced;
- the full local bond stress-slip relationship is obtained from the model, rather than just the maximum bond strength or reduction in anchorage capacity;
- reasonably good agreement with a large database of bond test results was shown.

Paper III. Partial safety factors for the anchorage capacity of corroded reinforcement bars in concrete.

In **Paper III**, partial safety factors were derived for the bond model established in **Paper II**. First, a probabilistic version of the bond model was set up by including the input parameters (basic variables) as random variables, following appropriate probability distributions. The response of the probabilistic model was studied using Monte Carlo simulations; over 600 design cases were run, with varied input for the basic variables such as concrete strength, embedment length and so on. Thereafter, probability distributions were fitted to the results and these were used to derive partial safety factors. In Figure 5.6 a), the resulting probability distributions (describing the anchorage capacity for three different embedment lengths) are shown for a sample case. Figure 5.6 b) shows the partial safety factors derived for 480 design cases with stirrups, in two different plots sorted by reinforcement bar diameter. It is worth noting that the smaller bar diameter required a higher partial factor for corrosion levels of 15 and 20%.



Figure 5.6. a) Probability distributions describing the anchorage capacity for three different embedment lengths for a sample case, plus the nominal, characteristic and design values. b) Comparison of partial factors for two different reinforcement bar diameters for the 480 design cases with stirrups. Adopted from **Paper III**.

The resulting partial safety factors, conservatively derived from the design cases, are assembled in Table 5.1.

Table 5.1. Partial safety factors for cases without and with stirrups.

	Corrosion level	Υм
Without stirrups	Uncorroded	2.0
	Corroded	3.4
With stirrups	0%	1.9
	5%	4.7
	10%	4.9
	15%	5.2-6.4*
	20%	5.2-7.6*

*Intermediate cases may be interpolated, in cases where n_b is 1-5.

The partial safety factors (shown in Table 5.1) were verified, based on reliability analyses of anchorages designed using the stated factors. The end region of a beam was considered, so that 20 design cases with stirrups might be composed, with varying embedment lengths, number of anchored bars and corrosion levels. Four design cases without stirrups were also included. A design value of the anchorage capacity was calculated for each design case, using the partial safety factors. By setting this value as equal to the design load, it was possible to determine the probability distributions of the loads for each design case. The reliability analyses were conducted by evaluating the limit-state function:

$$g(\mathbf{X}) = R_{\text{ARC2010}}(\mathbf{X}) - L(\mathbf{X})$$
(5.3)

where $R_{ARC2010}(\mathbf{X})$ is the anchorage capacity and $L(\mathbf{X})$ the load. For each design case, the resulting reliability was quantified using a combination of FORM and importance-sampling simulations. The resulting reliability indices and "sensitivity factors" for FORM are shown in Figure 5.7. Note that all reliability indices met the target reliability, β_t , of 3.7 for a one-year reference period. The sensitivity factors may be viewed as the influence of the basic variables on the reliability. The top four basic variables in the legend relate to the load, while the bottom four relate to the resistance. Note that with increasing corrosion level, the influence of the basic variable related to the load decreases, while the influence of the resistance-related basic variables increases. It is also noteworthy that the major influence of the resistance model uncertainty θ_R (which dominated for corrosion levels of 5%) successively decreases at higher corrosion levels, as the corrosion level itself exerts more influence.



Figure 5.7: a) resulting reliability indices for different corrosion levels, b) sensitivity factors obtained from FORM for a representative case with stirrups. Adopted from **Paper III**.

The main findings from Paper III may be summarised as follows:

- partial factors were derived for the anchorage capacity of corroded reinforcement bars in concrete;
- the intended safety level was verified using probabilistic reliability analyses for a number of designs, based on the partial safety factors;
- the modelling uncertainty was shown to be highly influential on the anchorage capacity of corroded reinforcement;
- the corrosion level was shown to influence the sensitivity factors of the basic variables.

Paper IV. Incorporation of pre-existing longitudinal cracks in finite element analyses of corroded reinforced concrete beams failing in anchorage.

Paper IV investigated methods of incorporating pre-existing, corrosion-induced, cracks in anchorage assessment. The methods were based on the four modelling levels proposed by [91], see Figure 5.8. **Paper IV** focused mainly on analyses on levels 3 and 4. 3D FEs were adopted for the concrete, which was modelled using a total-strain-based, smeared-crack approach. The reinforcement was modelled with 1D and 3D FEs for levels 3 and 4 respectively. Analysis on modelling level 3 depends on a bond stress-slip relation capable of representing the loss of confinement due to cracking of the concrete cover, cf. **Paper II**. Meanwhile, on level 4, the analyses use a friction model to explicitly include the confinement. Three methods of accounting for cracks were applied: a) modifying the bond stress-slip relation, b) weakening finite elements at the position of the crack and c) weakened discrete crack elements. All methods were applied on level 3, whereas b) and c) were applied on modelling level 4.



Figure 5.8. Overview of modelling levels for assessment of anchorage. Adopted from Paper IV as modified from [91].

Experiments on the anchorage capacity of corroded beam specimens from [92] were analysed using the three methods. An FE mesh of the concrete beam, load and support plates and the weakened and discrete crack elements are shown in Figure 5.9. Note that the reinforcement bars were de-bonded, except for the outer 100 mm on each side.

Resulting load-deflection curves for analyses on levels 3 and 4 are shown in Figure 5.10. The weakened elements or discrete cracks exerted no influence by on the results of level 3 analyses. Thus, this analysis level failed to describe the impact of pre-existing cracks. However, for level 4, the weakened elements led to a lower estimate of the ultimate capacity, compared to omitting cracks or implementing them as discrete cracks.



Figure 5.9. Elements assigned weakened elements on a) left-hand and b) right-hand side of specimen 5. c-e) Level of detail for the geometrical representation of discrete cracks for the left-hand side of the same specimen, with low, intermediate and high levels of detail shown in c-e) respectively. Adopted from **Paper IV**.



Figure 5.10. Load-deflection curves for specimen 5 analysed on a) level 3 and b) level 4, with crack implemented through the bond stress-slip relation (level 3) and through weakened elements (suffix Ip) and discrete cracks (suffix Id plus three levels Dl-h). From **Paper IV**.

The main findings from **Paper IV** may be summarised as follows:

- simple hand calculations might, with little effort, provide a lower bound for the capacity, without knowing the corrosion level;
- an increased modelling level resulted in improved prediction of the ultimate capacity, at the expense of increased analytical and computational efforts;
- pre-existing cracks in FE models with 1D reinforcement did not influence the results;
- the use of discrete cracks had no influence when included in the models with low, intermediate, or high detail, due to the stress state in the anchorage zone;
- 3D FEA with solid reinforcement elements (level 4) with cracks included as weakened elements provided the most complete and accurate results for this case. One advantage of this modelling method is that input regarding the corrosion level is not required.

Paper V. Incorporation of pre-existing cracks in finite element analyses of reinforced concrete beams prone to shear failure.

Paper V applied the method of incorporating pre-existing cracks in FE analyses (outlined for corrosion-induced cracks in **Paper IV**) to represent cracks from previous loading. A comparison was also made with experimental results from pre-cracked beams. Figure 3.7 presented the procedure for weakening the tensile properties of elements representing the cracks. The weakened properties of the discrete cracks were similarly derived (fully described in **Paper V**). Examples of pre-existing cracks incorporated using weakened elements were shown in Figure 3.8 and incorporation via discrete cracks is shown in Figure 5.11.



Figure 5.11. 3D mesh of PC1 with discrete cracks. Adopted from Paper V.

The load-deflection curves from FE analysis using weakened elements and discrete cracks, plus experimental results are shown in Figure 5.12.



Figure 5.12. Load-deflection curves for specimens PC1 to PC3, from experimental results and FE analyses of weakened elements (top row of plots), plus discrete cracks (bottom row of plots). The end of the analysis (unconverged step) is shown as a circle in each plot.

The main findings from **Paper V** may be summarised as follows:

- pre-existing cracks were shown to influence the failure mode, ultimate capacity and ductility. However, it was noted that the test setup was designed to make the cracks influential;
- incorporating pre-existing cracks in weakened elements led to better agreement with test results, compared to traditional FEA (without pre-existing cracks), particularly for the ductility and failure mode. However, the estimate of the ultimate capacity was also slightly improved;
- incorporation by discrete cracks gave improved prediction of the failure mode. However, traditional FEA provided better estimates of the ultimate capacity and ductility;
- modification of the compressive behaviour of the weakened elements (to reflect closure of cracks in the compressive zone), plus reduced reinforcement bond stiffness and strength (to represent damage from previous loading) was shown to influence the

bending stiffness in the analyses towards better correlation with the experimental observations;

• the choice of shear retention for the weakened elements was shown to influence the results from the analyses. However, its influence was minor for the discrete cracks.

6 Conclusions and future research

This section presents the conclusions drawn from the work within this thesis and offers suggestions for future research.

6.1 Conclusions

The aim of this thesis was to develop methods of reliable structural assessment of concrete structures with corrosion damage as well as pre-existing crack patterns. The work comprised methods of structural analysis and verification of structural safety. After brief introductory paragraphs, the main conclusions are presented in the same order as the research questions (see Section 1.2).

An investigation was conducted to see whether available safety formats for nonlinear analyses could also be used successfully on a more complicated geometry and loading (compared to a beam subjected to vertical load). The safety formats available in *fib* Model Code 2010 [16] were applied to a concrete frame, subjected to a vertical and horizontal (non-proportional) load. Two load histories were studied, a vertical load followed by a horizontal one and vice versa. Based on this investigation, the following conclusions could be drawn:

- the importance of load-history considerations in NLFE analysis was confirmed as, for one of the studied load histories, the ECOV safety format did not meet the intended safety level;
- a framework of engineering guidelines for NLFE analysis, dealing with all important aspects for conducting an accurate analysis, is desirable. Once this investigation was complete, guidelines were published by the Netherland's Ministry of Infrastructure and the Environment [20], which partly addressed this need. However, the author sees a need for clearer recommendations on how to treat the load history and how to define the design resistance for more complicated load cases.

Furthermore, a simplified model for assessing anchorage of corroded reinforcement in concrete was developed and validated (called ARC2010). This means that representative estimates may be obtained, of the anchorage capacity of corroded reinforcement in engineering applications. The following conclusions may be drawn based on the work with ARC2010:

- the model was developed based on understanding created by advanced models and experimental tests. It has the potential to assist practicing engineers in assessments, as it is straightforward to understand and use;
- despite the simplified nature of the model, the physical behaviour is well represented;
- the bond stress-slip relationships obtained from ARC2010 may also be used in NLFE analysis for a more realistic representation of the structural behaviour (regarding shear failure, for example);
- using *fib* Model Code 2010 as the basis for ARC2010 has made it possible to account for the reinforcement bar position in the cross-section and include other effects covered in the code, such as transverse stresses and longitudinal cracking;
- when compared to experimental results, ARC2010 predicts slightly lower normalised average bond strengths than those observed in the tests.

Moreover, verification of structural reliability in design and assessment typically compares the design value of the load effect with the design value of the resistance, cf. Eq. (4.4). Partial safety factors were derived for the proposed anchorage assessment model (ARC2020), to provide a design value for the anchorage capacity with a sufficient safety margin. Based on this derivation, the following conclusions may be drawn:

- a design value for anchorage capacity, consistent with the target reliability level, might be obtained by deriving partial safety factors for the anchorage capacity of corroded reinforcement in concrete;
- for assessing anchorage of corroded reinforcement, the modelling uncertainty is high. This contributes to high values for the partial factors.

There was also an examination of whether the influence of pre-existing cracks could be represented by locally weakening the material properties of the concrete. This was investigated for both continuum elements and discrete crack elements, in two markedly different situations: a) assessing the anchorage capacity, for corrosion-induced splitting cracks along the reinforcement and b) moment and shear capacity, for beams with cracks formed under previous external loading. Based on these investigations, the following conclusions may be drawn:

- an NLFE model with 3D elements for concrete and reinforcement (with splitting cracks due to reinforcement corrosion included as weakened continuum elements) provided a reasonable estimate of the anchorage capacity for the studied case, with no input regarding the corrosion level;
- the stress state in the anchorage zone meant no significant influence could be observed for the analyses of corrosion-induced cracks implemented using discrete cracks;
- when weakened continuum elements were used to incorporate pre-existing cracks in FE models of RC beams (with cracks from previous external load tested in bending), improved estimates (compared to traditional FEA without pre-existing cracks) were obtained, particularly for the ductility and failure mode. However, the ultimate capacity was also slightly improved;
- incorporation of pre-existing cracks by discrete crack elements for the same case did not improve estimates of the ultimate capacity and ductility compared to traditional FEA. However, the failure mode was better estimated;
- compared to discrete cracks, less implementation effort and computational time was required for crack incorporation using weakened continuum elements.

Finally, some overarching conclusions addressing the aim of the work may be drawn from the aggregated work. These may be summarised as follows:

- if improved models for estimating the capacity of damaged structures are applied more frequently in practice, unnecessary strengthening or dismantling of structures may be reduced;
- when simplified models are not sufficient, perhaps because they have been simplified, or because there is a requirement for input data which is not available (such as corrosion level), advanced models for improved capacity estimates may be used but at the cost of greater effort;
- to be practically relevant, simplified and advanced models must be equipped with suitable safety formats, if they are to produce results with a sufficient safety margin.

6.2 Suggestions for future research

Based on the work conducted in this thesis, multiple areas of importance for future work have been identified. The following suggestions may be offered:

- in the evaluation of safety formats for NLFE analysis, it was found that one safety format did not lead to the intended safety level. It would be interesting to further study the applicability of available safety formats for cases with more complex loading and geometry (such as a shell structure subjected to a variety of loading conditions). Different failure modes and their effect on the modelling uncertainty should be included;
- in the calibration of partial safety factors conducted in this work, it was evident that the common assumption for the sensitivity factor for resistance ($\alpha_R = 0.8$) did not correspond well at higher corrosion levels. A study of sensitivity factors for basic variables for deteriorated concrete structures is therefore proposed;
- many of the assessment methods for concrete structures with corroded reinforcement require input regarding the corrosion level. For now, this is often difficult to determine in practice. Development of non or semi-destructive methods to estimate the corrosion level of reinforcement bars in existing concrete structures would therefore be of major practical importance.

References

- [1] European Conference of Ministers of Transport. Investment in Transport Infrastructure 1985-1995. Paris, France: OECD Publishing; 1999.
- [2] Melchers RE. Assessment of existing structures approaches and research needs. J Struct Eng 2001;127:406–11. https://doi.org/https://doi.org/10.1061/(ASCE)0733-9445(2001)127:4(406).
- [3] Hobbs DW. Concrete deterioration: Causes, diagnosis, and minimising risk. Int Mater Rev 2001;46:117–44. https://doi.org/10.1179/095066001101528420.
- [4] Cairns J, Du Y, Law D. Structural performance of corrosion-damaged concrete beams. Mag Concr Res 2008;60:359–70.
- [5] CEB-fib. fib Bulletin 10 Bond of reinforcement in concrete. Lausanne, Switzerland: International Federation for Structural Concrete; 2000.
- [6] Almusallam AA, Al-Gahtani AS, Aziz AR, Rasheeduzzafar A. Effects of reinforcement corrosion on bond strength. Constr Build Mater 1996;10:123–9.
- [7] Al-Sulaimani GJ, Kaleemullah M, Basunbul IA, Rasheeduzzafar. Influence of corrosion and cracking on bond behavior and strength of reinforced concrete members. ACI Struct J 1990;87:220–31.
- [8] Mangat PS, Elgarf MS. Bond characteristics of corroding reinforcement in concrete beams. Mater Struct 1999;32:89–97.
- [9] Tahershamsi M, Zandi K, Lundgren K, Plos M. Anchorage of naturally corroded bars in reinforced concrete structures. Mag Concr Res 2014;66:729–44. https://doi.org/10.1680/macr.13.00276.
- [10] Mancini G, Tondolo F. Effect of bond degradation due to corrosion A literature survey. Struct Concr 2014;15:408–18.
- [11] Lundgren K. Bond between ribbed bars and concrete. Part 2: The effect of corrosion. Mag Concr Res 2005;57:383–95.
- [12] Ožbolt J, Oršanić F, Balabanić G. Modelling processes related to corrosion of reinforcement in concrete: coupled 3D finite element model. Struct Infrastruct Eng 2017;13:135–46. https://doi.org/10.1080/15732479.2016.1198400.
- [13] Zandi Hanjari K, Lundgren K, Plos M, Coronelli D. Three-dimensional modelling of structural effects of corroding steel reinforcement in concrete. Struct Infrastruct Eng 2013;9:702–18.
- [14] ISO13822. Basis for design of structures Assessment of existing structures. Geneva, Switzerland: ISO; 2001.
- [15] CEN. EN 1990:2002 Eurocode Basis of structural design. Brussels: European Committee for Standardization; 2002.
- [16] FIB. Model Code 2010. Lausanne, Switzerland: 2013.
- [17] Engen M, Hendriks MAN, Øverli JA, Åldstedt E. Solution strategy for non-linear finite element analyses of large reinforced concrete structures. Struct Concr 2015;16:389–97. https://doi.org/10.1002/suco.201400088.
- [18] Engen M, Hendriks MAN, Köhler J, Øverli JA, Åldstedt E. A quantification of the

modelling uncertainty of non-linear finite element analyses of large concrete structures. Struct Saf 2017;64:1–8. https://doi.org/10.1016/j.strusafe.2016.08.003.

- [19] Schlune H, Plos M, Gylltoft K. Safety formats for non-linear analysis of concrete structures. Mag Concr Res 2012;64:563-574(11).
- [20] Hendriks MAN, de Boer A, Belletti B. Guidelines for Nonlinear Finite Element Analysis of Concrete Structures. Report RTD:1016-1:2017: 2017.
- [21] Cervenka J, Cervenka V, Sykora M, Mlcoch J. Evaluation of safety formats for structural assessment based on nonlinear analysis. In: Meschke G, Pichler B, Rots JG, editors. Comput. Model. Concr. Struct., London: CRC Press; 2018.
- [22] Holický M, Návarová V, Gottfried R, Kronika M, Marková J, Miroslav Sýkora KJ. Basics for assessment of existing structures. 1st ed. Prague: Klokner Institute, Czech Technical University in Prague; 2013.
- [23] Larosche CJ. Types and causes of cracking in concrete structures. vol. c. Woodhead Publishing Limited; 2009. https://doi.org/10.1533/9781845697037.1.57.
- [24] for Standardization EC. EN 1992-1-1 Eurocode 2: Design of concrete structures Part 1-1: General rules and rules for buildings, Brussels: CEN; 2005.
- [25] Tan R, Eileraas K, Kanstad T, Opkvitne O, Žirgulis G, Hendriks MAN, et al. Experimental and theoretical investigation of crack width calculation methods for RC ties. Struct Concr 2018;19:1436–47. https://doi.org/10.1002/suco.201700237.
- [26] Wang J, Shi Z, Nakano M. Strength degradation analysis of an aging RC girder bridge using FE crack analysis and simple capacity-evaluation equations. Eng Fract Mech 2013;108:209–21. https://doi.org/10.1016/j.engfracmech.2013.04.011.
- [27] Bradley D, Hehenberger P. Mechatronic Futures. Mechatron. Futur., Cham: Springer International Publishing; 2016, p. 1–15. https://doi.org/10.1007/978-3-319-32156-1_1.
- [28] Zandi K, Ransom EH, Topac T, Chen R, Beniwal S, Blomfors M, et al. A Framework For Digital Twin of Civil Infrastructure - Challenges and Opportunities. 12th Int. Work. Struct. Heal. Monit. Stanford, California, USA, Sept. 10-12, 2019, Lancaster, PA, USA: DEStech Publications, Inc.; 2019, p. 7 pp.
- [29] Kim H, Lee J, Ahn E, Cho S, Shin M, Sim SH. Concrete crack identification using a UAV incorporating hybrid image processing. Sensors 2017;17:1–14. https://doi.org/10.3390/s17092052.
- [30] Kim IH, Jeon H, Baek SC, Hong WH, Jung HJ. Application of crack identification techniques for an aging concrete bridge inspection using an unmanned aerial vehicle. Sensors 2018;18:1–14. https://doi.org/10.3390/s18061881.
- [31] Kapoor M, Katsanos E, Thöns S, Nalpantidis L, Winkler J. Structural integrity management with unmanned aerial vehicles: State-of-the-art review and outlook. In: Caspeele R, Taerwe L, Frangopol DM, editors. Life-Cycle Anal. Assess. Civ. Eng. Towar. an Integr. Vis. - Proc. 6th Int. Symp. Life-Cycle Civ. Eng. IALCCE 2018, London, UK: Taylor & Francis; 2019, p. 2161–8.
- [32] Berrocal CG, Fernandez I, Rempling R. Crack monitoring in reinforced concrete beams by distributed optical fiber sensors. Struct Infrastruct Eng 2020;0:1–16. https://doi.org/10.1080/15732479.2020.1731558.
- [33] Brault A, Hoult NA, Greenough T, Trudeau I. Monitoring of Beams in an RC Building during a Load Test Using Distributed Sensors. J Perform Constr Facil 2019;33:1–11.

https://doi.org/10.1061/(ASCE)CF.1943-5509.0001250.

- [34] ISO2394. General principles on reliability of structures. Geneva, Switzerland: ISO; 1998.
- [35] ISO12491. Statistical methods for quality control of building materials and components. Geneva, Switzerland: ISO; 1998.
- [36] Bell B. Sustainable bridges. D1.3 European Railway Bridge Problems. Report: 2004.
- [37] Caspeele R, Sykora M, Allaix DL, Steenbergen R. The design value method and Adjusted Partial Factor Approach for existing structures. Struct Eng Int J Int Assoc Bridg Struct Eng 2013;23:386–93. https://doi.org/10.2749/101686613X13627347100194.
- [38] Cook RD, Malkus DS, Plesha ME, Witt RJ. Concepts and Applications of Finite Element Analysis. Fourth edi. New York, USA: John Wiley & Sons; 2002.
- [39] Licko. Wikipedia Commons 2007. https://commons.wikimedia.org/wiki/File:Hoover_dam_from_air.jpg.
- [40] Hinton E. NAFEMS Introduction to Nonlinear Finite Element Analysis. NAFEMS; 1992.
- [41] Pacoste C, Plos M, Johansson M. Recommendations for finite element analysis for design of reinforced concrete bridges. TRITA-BKN Rapport 144. Stockholm, Sweden: 2012.
- [42] TNO DIANA. DIANA 10.3 Users manual 2019.
- [43] C.L. Page, K. W. J. Treadaway. Aspects of electrochemistry of steel in concrete. Nature 1982;5862:109–15.
- [44] Ghods P. Multiscale Investigation of the Formation and Breakdown of Passive Films on Carbon Steel Rebar in Concrete. Carleton University, 2010. https://doi.org/10.1149/ma2010-02/13/1204.
- [45] Bertolini L, Elsener B, Pedeferri P, Redaelli E, Polder RB. Corrosion of Steel in Concrete: Prevention, Diagnosis, Repair. Second. Weinheim, Germany: John Wiley & Sons; 2013.
- [46] Morinaga S. Remaining life of reinforced concrete structures after corrosion cracking. In: Sjöström C, editor. Durab. Build. Mater. components 7 (volume one), London, UK: Taylor & Francis; 1996, p. 685. https://doi.org/10.4324/9781315025018.
- [47] Riskin J, Khentov A. Electrocorrosion and protection of metals. Second edi. Amsterdam, Netherlands: Elsevier; 2019.
- [48] Bentur A, Berke N, Diamond S. Steel Corrosion in Concrete Fundamentals and civil engineering practice. London, UK: CRC Press; 1997. https://doi.org/10.1201/9781482271898.
- [49] Andrade C, Alonso C. Test methods for on-site corrosion rate measurement of steel reinforcement in concrete by means of the polarization resistance method. Mater Struct 2004;37:623–43. https://doi.org/10.1007/BF02483292.
- [50] Tang L, Malmberg B. Assessment of reinforcement corrosion in a concrete highway tunnel. In: Alexander M, Beushausen H-D, Dehn F, Moyo P, editors. Concr. Repair, Rehabil. Retrofit., London, UK: Taylor & Francis; 2006.
- [51] Khan I, François R, Castel A. Prediction of reinforcement corrosion using corrosion induced cracks width in corroded reinforced concrete beams. Cem Concr Res 2014;56:84–96. https://doi.org/10.1016/j.cemconres.2013.11.006.
- [52] Coronelli D, Gambarova P. Structural Assessment of Corroded Reinforced Concrete

Beams: Modeling Guidelines. J Struct Eng 2004;130:1214–24. https://doi.org/10.1061/(ASCE)0733-9445(2004)130:8(1214).

- [53] Clark L a., Chan a. HC, Du YG. Residual capacity of corroded reinforcing bars. Mag Concr Res 2005;57:135–47. https://doi.org/10.1680/macr.2005.57.3.135.
- [54] Chan a. HC, Clark L a., Du YG. Effect of corrosion on ductility of reinforcing bars. Mag Concr Res 2005;57:407–19.
- [55] Auyeung Y, Balaguru P, Chung L. Bond Behavior of Corroded Reinforcement Bars. ACI Mater J 2000;97:214–20.
- [56] Cabrera, J.G. and Ghoddousi P. The effect of reinforcement corrosion on the srength of the steel concrete interface. Int Conf Bond Concr from Res to Pract Proceedings, 1992:11–24.
- [57] Al-Sulaimani GJ, Kaleemullah M, Basunnul IA, Rasheeduzzafar AA-M. Influence of corrosion and cracking on bond behaviour and strength of reinforced concrete members. ACI Struct J 1990;97:220–31.
- [58] Rodriguez J, Ortega L, Casal J. Load carrying capacity of concrete structures with corroded reinforcement. Constr Build Mater 1997;11:239–48.
- [59] Berra M, Castellani A, Coronelli D, Zanni S, Zhang G. Steel–concrete bond deterioration due to corrosion: finite-element analysis for different confinement levels. Mag Concr Res 2003;55:237–47. https://doi.org/10.1680/macr.55.3.237.37574.
- [60] Fernandez I, Bairán JM, Marí AR. Corrosion effects on the mechanical properties of reinforcing steel bars. Fatigue and sigma-epsilon behavior. Constr Build Mater 2015;101:772–83.
- [61] Chen E, Berrocal CG, Fernandez I, Löfgren I, Lundgren K. Assessment of the mechanical behaviour of reinforcement bars with localised pitting corrosion by Digital Image Correlation. Eng Struct 2020;219:1–17. https://doi.org/10.1016/j.engstruct.2020.110936.
- [62] Vecchio FJ, Collins MP. The Modified Compression-Field Theory for Reinforced Concrete Elements Subjected to Shear. ACI J 1986;83:219–31. https://doi.org/10.14359/10416.
- [63] Castel A, Khan I, Francois R, Gilbert RI. Modeling steel concrete bond strength reduction due to corrosion. ACI Struct J 2016;113:973–82.
- [64] Prieto M, Tanner P, Andrade C. Multiple linear regression model for the assessment of bond strength in corroded and non-corroded steel bars in structural concrete. Mater Struct 2016;49:4749–63.
- [65] Lee H-S, Noguchi T, Tomosawa F. Evaluation of the bond properties between concrete and reinforcement as a function of the degree of reinforcement corrosion. Cem Concr Res 2002;32:1313–8.
- [66] Zandi K. Corrosion-induced cover spalling and anchorage capacity. Struct Infrastruct Eng 2015;11:1–18.
- [67] CEN. SS-EN 1992-1-1:2005. 2008.
- [68] Concrete Society. Non-structural cracks in concrete: a Concrete Society report. Technical Report No. 22 - Fourth Edition. London, UK: 2010.
- [69] Nordström E, Hassanzadeh M, Malm R, Ekström T, Janz M. Structural safety of cracked

concrete dams. Report 2019:623. 2019.

- [70] Rots JG, Blaauwendraad J. Crack models for concrete: discrete or smeared? Fixed multidirectional or rotating? Heron 1989;34:3–59. https://doi.org/10.1096/fj.02.
- [71] Collins MP, Mitchell D. Prestressed concrete structures. Englewood Cliffs, N.J.: Prentice Hall; 1991.
- [72] Toutlemonde F, Pascu R, Ranc G, Kretz T. Analysis of the structural effects of throughcracking within a reinforced concrete slab. Bull Des Lab Des Ponts Chaussées 2005:193– 213.
- [73] Talley KG, Arrellaga J, Breen JE. Computational modeling of existing damage in concrete bridge columns. J Struct Eng 2014;140:1–6. https://doi.org/10.1061/(ASCE)ST.1943-541X.0001115.
- [74] Dhanmeher S. Crack pattern observations to finite element simulation. Delft University of Technology, 2017.
- [75] Hanjari KZ, Kettil P, Lundgren K. Modelling the structural behaviour of frost-damaged reinforced concrete structures. Struct Infrastruct Eng 2013;9:416–31. https://doi.org/10.1080/15732479.2011.552916.
- [76] JCSS. Probabilistic Model Code, 2001.
- [77] Melchers RE, Beck AT. Structural Reliability Analysis and Prediction. Third edit. Hoboken, New Jersey, USA: John Wiley & Sons; 2018.
- [78] Schlune H. Safety Evaluation of Concrete Structures with Nonlinear Analysis. Institutionen för bygg- och miljöteknik, Konstruktionsteknik, Chalmers tekniska högskola,; 2011.
- [79] Marelli S, Schöbi R, Sudret B. UQLab user manual- structural reliability (rare event estimation). 2019.
- [80] Marelli S, Sudret B. UQLab: A Framework for Uncertainty Quantification in Matlab. 2nd Int. Conf. Vulnerability Risk Anal. Manag. (ICVRAM 2014), Reston, VA: American Society of Civil Engineers; 2014, p. 2554–63. https://doi.org/10.1061/9780784413609.257.
- [81] Ditlevsen O, Madsen HO. Structural Reliability Methods. Chichester: John Wiley & Sons Ltd; 1996.
- [82] El-Reedy MA. Reinforced Concrete Structural Reliability. Boca Raton: CRC Press; 2012. https://doi.org/10.1201/b12978.
- [83] Hasofer a M, Lind NC. An exact and invariant first order reliability format. J Eng Mech Div ASCE 1974;100:111–21.
- [84] Rackwitz R, Flessler B. Structural reliability under combined random load sequences. Comput Struct 1978;9:489–94. https://doi.org/10.1016/0045-7949(78)90046-9.
- [85] Zhang Y, Der Kiureghian A. Two improved algorithms for reliability analysis. In: Rackwitz R, Augusti G, Bori A, editors. Proc. 6th IFIP WG7.5 Reliab. Optim. Struct. Syst., London: Chapman & Hall; 1995, p. 297–304.
- [86] Olsson A, Sandberg G, Dahlbom O. On Latin hypercube sampling for structural reliability analysis. Struct Saf 2003:47–68.
- [87] Mann NR, Schafer RE, Singpurwalla ND. Methods for Statistical Analysis of Reliability

and Life Data. New York: JohnWiley; 1974.

- [88] Rubinstein RY. Simulation and the Monte Carlo Method. New York: JohnWiley & Sons; 1981.
- [89] Bucher CG, Bourgund U. A fast and efficient response surface approach for structural reliability problems. Struct Saf 1990;7:57–66. https://doi.org/10.1016/0167-4730(90)90012-E.
- [90] Rajashekhar MR, Ellingwood BR. A new look at the response surface approach for reliability analysis. Struct Saf 1993;12:205–20.
- [91] Tahershamsi M, Fernandez I, Zandi K, Lundgren K. Four levels to assess anchorage capacity of corroded reinforcement in concrete. Eng Struct 2017;147:434–47. https://doi.org/10.1016/j.engstruct.2017.06.024.
- [92] Zandi K, Boubitsas D, Fahimi S, Johansson M, Spetz J, Flansbjer M. Autonomous Automated Non-Intrusive Condition Assessment of Concrete Structures. Gothenburg. Report ACE 2019:5: 2019.

Bibliography, Concrete Structures

Theses are given in order of time of publication. The numbers refer to the two series of dissertations at Chalmers University of Technology (the original, 1941-1975, and the new from 1971). The first PhD dissertation was arranged in 1948 although the first theses were published earlier. Theses until 1966 are in Civil Engineering, while theses from 1971 are in Concrete Structures.

Doctoral Theses

- [1] Bertil Löfquist. "Temperatureffekter i hårdnande betong. Undersökning av några faktorer som påverka sprickbildningen i grövre konstruktioner—Jämförelse mellan två svenska cement". No. 3, 195 pp (in Swedish). PhD thesis. Chalmers University of Technology, 1946.
- [2] Nils Tengvik. "Den svenska byggnadsmaterialmarknaden. Produktion, distribution och prissättning av jord- och stenindustrins material". No. 7, 318 pp. (in Swedish). PhD thesis. Chalmers University of Technology, 1952.
- [3] Sven Ödman. "Studies of Boundary Value Problems". No. 10, 10 pp + app. PhD thesis. Chalmers University of Technology, 1955.
- [4] Gunnar Kärrholm. "Parallelogram Plates Analysed by Strip Method". No. 11, 196 pp. PhD thesis. Chalmers University of Technology, 1956.
- [5] Lars-Erik Larsson. "Bearing Capacity of Plain and Reinforced Concrete Walls". No.20, 248 pp. PhD thesis. Chalmers University of Technology, 1959.
- [6] Sven Hansbo. "Consolidation of Clay, with Special Reference to Influence of VerticalSand Drains. A Study Made in Connection with Full-scale Investigation at Skå-Edeby". No. 25, 160 pp. PhD thesis. Chalmers University of Technology, 1960.
- [7] Per Anders Hedar. "Stability of Rock-fill Breakwaters". No 26, 119 pp. PhD thesis. Chalmers University of Technology, 1960.
- [8] Anders Losberg. "Structurally Reinforced Concrete Pavements." No. 29, 444 pp. PhD thesis. Chalmers University of Technology, 1960.
- [9] Alf Samuelsson. "Linear Analysis of Frame Structures by Use of Algebraic Topology". No. 36, 115 pp. PhD thesis. Chalmers University of Technology, 1962.
- [10] Bengt Åkesson. "On Lévy's Plate Solution". No. 36, 115 pp. PhD thesis. Chalmers University of Technology, 1966.15
- [11] Gunnar Backsell. "Experimental Investigations into Deformations Resulting from Stresses Perpendicular to Grain in Swedish Whitewood and Redwood in Respect of the Dimensioning of Concrete Formwork". No. 57, 113 pp. PhD thesis. Chalmers University of Technology, 1966.
- [12] Krister Cederwall. "Time-dependent behaviour of reinforced concrete structures". No109. Publication 71:3 by the National Swedish Building Research and as Publication71:1 by the Division of Concrete Structures, 173 pp. PhD thesis. Chalmers University of Technology, 1971.

- [13] Lennart Elfgren. "Reinforced Concrete Beams Loaded in Combined Torsion, Bending and Shear: A Study of the Ultimate Load-Carrying Capacity". New Series No 28, Publication 71:3, 230 pp., Second Ed 1972, 230 pp. PhD thesis. Chalmers University of Technology, 1971.
- [14] Inge Karlsson. "Stiffness Properties of Reinforced Concrete Beams in Combined Torsion, Bending and Shear". No 126, Publication 73:1, 203 pp. PhD thesis. Chalmers University of Technology, 1973.
- [15] Ingvar H. E. Nilsson. "Reinforced Concrete Corners and Joints Subjected to Bending Moment: Design of Corners and Joints in Frame Structures". New Series No 90.Also published as Document 1973:7 by the National Swedish Building Research and as Publication 73:6 by the Division of Concrete Structures, 249 pp. PhD thesis. Chalmers University of Technology, 1973.
- [16] Ralejs Tepfers. "A Theory of Bond Applied to Overlapped Tensile Reinforcement Splices for Deformed Bars". Second edition 1976. No. 732, Publication 73:2, 328 pp. PhD thesis. Chalmers University of Technology, 1973.
- [17] Arne Cajdert. "Laterally Loaded Masonry Walls". New Series No 346, No 865, Publication 80:5, 283 pp. PhD thesis. Chalmers University of Technology, 1980.
- [18] Olav Berge. "Armerade konstruktioner i lättballastbetong". New Series No 908. Publication 81:3, 373 pp. (In Swedish). PhD thesis. Chalmers University of Technology, 1981.
- [19] Stig Öberg. "Post Tensioned Shear Reinforcement in Rectangular RC Beams". No.1021, Publication 90:1, 603 pp. PhD thesis. Chalmers University of Technology, 1990.
- [20] Björn Engström. "Ductility of Tie Connections in Precast Structures". Nos. 936, 999, 1023, 1052, Publication 92:1, 368 pp. + app. PhD thesis. Chalmers University of Technology, 1992.
- [21] Marianne Grauers. "Composite Columns of Hollow Steel Sections Filled with High Strength Concrete". No. 1077, Publication 93:2, 140 pp. PhD thesis. Chalmers University of Technology, 1993.
- [22] Li An. "Load Bearing Capacity and Behaviour of Composite Slabs with Profiled Steel Sheet". No. 1075, Publication 93:4, 134 pp. PhD thesis. Chalmers University of Technology, 1993.
- [23] Mario Plos. "Application of Fracture Mechanics to Concrete Bridges. Finite Element Analysis and Experiments". Nos. 1067, 1084, 1088, 1106, Publication 95:3, 127 pp. PhD thesis. Chalmers University of Technology, 1995.
- [24] Magnus Åkesson. "Implementation and Application of Fracture Mechanics Models for Concrete Structures". No 1112, Publication 96:2, 159 pp. PhD thesis. Chalmers University of Technology, 1996.
- [25] Christina Claeson. "Structural Behavior of Reinforced High-Strength Concrete Columns". No 1105, Publication 98:1, 92 pp. PhD thesis. Chalmers University of Technology, 1998.
- [26] Karin Lundgren. "Three-Dimensional Modelling of Bond in Reinforced Concrete. Theoretical Model, Experiments and Applications". No 37, Publication 99:1, 129 pp. PhD thesis. Chalmers University of Technology, 1999.

- [27] Jonas Magnusson. "Bond and Anchorage of Ribbed Bars in High-Strength Concrete". No 1113, Publication 00:1, 300 pp. PhD thesis. Chalmers University of Technology, 2000.
- [28] Morgan Johansson. "Structural Behaviour in Concrete Frame Corners of Civil De-fence Shelters". No 1106, Publication 00:2, 220 pp. PhD thesis. Chalmers University of Technology, 2000.
- [29] Rikard Gustavson. "Structural Behaviour of Concrete Railway Sleepers". No 32, Publication 02:6, 180 pp. PhD thesis. Chalmers University of Technology, 2002.
- [30] Mathias Johansson. "Composite Action and Confinement Effects in Tubular Steel-Concrete Columns". No 33, Publication 02:8, 173 pp. PhD thesis. Chalmers University of Technology, 2002.
- [31] Peter Grassl. "Plasticity and Damage Mechanics for Modeling Concrete Failure". Publication 04:3, 159 pp. PhD thesis. Chalmers University of Technology, 2004.
- [32] Joosef Leppänen. "Concrete Structures Subjected to Fragment Impacts—Dynamic Behaviour and Material Modelling". No. 31, Publication 04:4, 125 pp. PhD thesis. Chalmers University of Technology, 2004.
- [33] Ingemar Löfgren. "Fibre-reinforced Concrete for Industrial Construction a fracture mechanics approach to material testing and structural analysis". Ny serie nr 2378, 243 pp. PhD thesis. Chalmers University of Technology, 2005.
- [34] Per-Ola Svahn. "Dynamic Behaviour of Reinforced Concrete Structures: Analyses with a Strong Discontinuity Approach". Ny serie nr 2366, 159 pp. PhD thesis. Chalmers University of Technology, 2005.17
- [35] Helen Broo. "Shear and Torsion in Concrete Structures. Non-Linear Finite Element Analysis in Design and Assessment". Ny serie nr 2786. PhD thesis. Chalmers University of Technology, 2008.
- [36] Peter Harryson. "Industrial Bridge Engineering—Structural developments for more efficient bridge construction". Ny serie nr 2810. PhD thesis. Chalmers University of Technology, 2008.
- [37] Rasmus Rempling. "Modelling of Concrete Subjected to Cyclic Loading". Ny serie nr 2927, 169 pp. PhD thesis. Chalmers University of Technology, 2009.
- [38] Kamyab Zandi Hanjari. "Structural Behaviour of Deteriorated Concrete Structures". Ny serie nr 3142, 64 pp. PhD thesis. Chalmers University of Technology, 2010.
- [39] Anette Jansson. "Effects of Steel Fibres on Cracking in Reinforced Concrete". Ny serie nr 3233, 50 pp. PhD thesis. Chalmers University of Technology, 2011.
- [40] Hendrik Schlune. "Safety Evaluation of Concrete Structures with Nonlinear Analysis". Ny serie nr 3232, 45 pp. PhD thesis. Chalmers University of Technology, 2011.
- [41] Ulrika Nyström. "Modelling of Concrete Structures Subjected to Blast and Fragment Loading". New Series No 3486, 83 pp. + papers. PhD thesis. Chalmers University of Technology, 2013.
- [42] Filip Nilenius. "Moisture and Chloride Transport in Concrete: Mesoscale Modelling and Computational Homogenization". New Series No 3658, 29 pp. + papers. PhD thesis. Chalmers University of Technology, 2014.

- [43] David Fall. "Steel Fibres in Reinforced Concrete Structures of Complex Shapes: Structural Behaviour and Design Perspectives." New Series No 3699, 68 pp. + papers. PhD thesis. Chalmers University of Technology, 2014.
- [44] Natalie Williams Portal. "Usability of Textile Reinforced Concrete: Structural Performance, Durability and Sustainability". New Series No 3914, 115 pp. + papers. PhD thesis. Chalmers University of Technology, 2015.
- [45] Mohammad Tahershamsi. "Structural Effects of Reinforcement Corrosion in Concrete Structures". New Series No 4108, 40 pp. + papers. PhD thesis. Chalmers University of Technology, 2016.
- [46] Jiangpeng Shu. "Structural Analysis Methods for Assessment of Reinforced Concrete Slabs". New Series No 4265, 54 pp. + papers. PhD thesis. Chalmers University of Technology, 2017.
- [47] Carlos Gil Berrocal. "Corrosion of Steel Bars in Fibre Reinforced Concrete: Corrosion mechanisms and structural performance". New Series No 4289, 69 pp. + papers. PhD thesis. Chalmers University of Technology, 2017.18
- [48] Jonas Ekström. "Blast and Impact Loaded Concrete Structures: Numerical and Experimental Methodologies for Reinforced Plain and Fibre Concrete Structures". New Series No 4325, 30 pp. + papers. PhD thesis. Chalmers University of Technology, 2017.
- [49] Daniel Ekström. "Integrated structural and construction engineering A study of project team performance in Swedish bridge design". New Series No 4655, 43 pp. + papers. PhD thesis. Chalmers University of Technology, 2019.
- [50] Adam Sciegaj. "Multiscale Modelling of Reinforced Concrete Structures". New Series No 4705, 45 pp. + papers. PhD thesis. Chalmers University of Technology, 2020.

Licentiate Theses

- [1] Ralejs Tepfers. "Studium av förankrings och sprickproblem hos balkar armerade med Ks 60". No. 626, 99 pp. Licentiate Thesis (part I). Chalmers University of Technology, 1962.
- [2] Ralejs Tepfers. "Problem vid avkortning av dragarmering: Föredrag vid Smedjebackens Valsverks Byggdagar". No. 639, 24 pp. Licentiate Thesis (part II). Chalmers University of Technology, 1963.
- [3] Ralejs Tepfers. "Försök med hålbalkar". No. 66:1. Licentiate Thesis (part III). Chalmers University of Technology, 1966.
- [4] Kent Gustavsson. "Fogar i samverkande betongkonstruktioner med tunn pågjutning". No. 938, Publication 81:7, 111 pp. Licentiate Thesis. Chalmers University of Technology, 1981.
- [5] Krister Cederwall and Thomas Pettersson. "Spännarmerad betong Tidsberoende spännförluster". No. 1053, Publication 85:5, 36 pp. Licentiate Thesis. Chalmers University of Technology, 1985.
- [6] Thomas Pettersson. "Tvärkraftskapacitet hos förspända balkar i snitt med stort moment". No. 1004, Publication 86:1, 45 pp. Licentiate Thesis. Chalmers University of Technology, 1986.
- [7] Steve Svensson. "Skivverkan i elementbjälklag. Skjuvkraftsöverföring i uppspruckna fogar". No. 1026, Publication 88:1, 59 pp. Licentiate Thesis. Chalmers University of Technology, 1988.

- [8] Elzbieta Saran. "Shear Capacity of Composite Prestressed Concrete Beams". No.1081, Publication 89:1, 127 pp. Licentiate Thesis. Chalmers University of Technology, 1989.
- [9] Johannes Svedin. "En modell for beräkning av sprickbredd i armerade betongkonstruktioner påverkade av statisk engångslast". No. 1083, Publication 89:2, 95 pp. Licentiate Thesis. Chalmers University of Technology, 1989.
- [10] Marianne Grauers. "Shear Capacity of Three-Layered Composite Concrete Slabs". No.1065, Publication 89:3, 27 pp. + Appendix. Licentiate Thesis. Chalmers University of Technology, 1989.
- [11] Johan Hedin. "Långtidsegenskaper hos samverkanskonstruktioner av stål och be-tong". No. 1079, Publication 90:2, 53 pp. (In Swedish). Licentiate Thesis. ChalmersUniversity of Technology, 1990.
- [12] Mario Plos. "Shear Behaviour in Concrete Bridges Full Scale Shear Test. Fracture Mechanics Analyses and Evaluation of Code Model". Nos. 1088, 1084, Publication 93:1, 70 pp. Licentiate Thesis. Chalmers University of Technology, 1993.
- [13] Magnus Åkesson. "Fracture Mechanics Analysis of the Transmission in Zone in Prestressed Hollow Core Slabs". No. 1112, Publication 93:5, 64 pp. Licentiate Thesis. Chalmers University of Technology, 1993.
- [14] Christina Claeson. "Behavior of Reinforced High Strength Concrete Columns". No.1105, Publication 95:1, 54 pp. Licentiate Thesis. Chalmers University of Technology, 1995.
- [15] Karin Lundgren. "Slender Precast Systems with Load-Bearing Facades". No. 1098, Publication 95:2, 60 pp. Licentiate Thesis. Chalmers University of Technology, 1995.
- [16] Morgan Johansson. "New Reinforcement Detailing in Concrete Frame Corners of Civil Shelters. Non-linear Finite Element Analyses and Experiments". No. 1106, Publication 96:1, 77 pp. Licentiate Thesis. Chalmers University of Technology, 1996.
- [17] Jonas Magnusson. "Bond and Anchorage of Deformed Bars in High-Strength Concrete". No. 1113, Publication 97:1, 234 pp. Licentiate Thesis. Chalmers University of Technology, 1997.
- [18] Rikard Gustavson. "Static and Dynamic Finite Element Analyses of Concrete Sleepers". No. 41, Publication 00:3, 58 pp. Licentiate Thesis. Chalmers University of Technology, 2000.
- [19] Mathias Johansson. "Structural Behaviour of Circular Steel-Concrete Columns. Nonlinear Finite Element Analyses and Experiments". No. 48, Publication 00:4, 64 pp. Licentiate Thesis. Chalmers University of Technology, 2000.
- [20] Gunnar Holmberg. "Fatigue of Concrete Piles of High Strength Concrete Exposed to Impact Load". No. 55, Publication 01:3, 69 pp. Licentiate Thesis. Chalmers University of Technology, 2001.
- [21] Peter Harryson. "Industrial Bridge Construction—merging developments of process, productivity and products with technical solutions". No. 34, Publication 02:1, 90 pp. Licentiate Thesis. Chalmers University of Technology, 2002.
- [22] Ingemar Löfgren. "In-situ concrete building systems developments for industrial constructions". No. 35, Publication 02:2, 125 pp. Licentiate Thesis. Chalmers University of Technology, 2002.

- [23] Joosef Leppänen. "Dynamic Behaviour of Concrete Structures subjected to Blast and Fragment Impacts". No. 31, Publication 02:4, 78 pp. Licentiate Thesis. Chalmers University of Technology, 2002.
- [24] Peter Grassl. "Constitutive Modelling of Concrete in Compression". No. 37, Publication 02:4, 95 pp. Licentiate Thesis. Chalmers University of Technology, 2002.
- [25] Per-Ola Svahn. "Impact Loaded Concrete Piles—Theoretical and experimental study of load effects and capacity". No 38, Publication 03:1, 51 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2003.
- [26] Helen Broo. "Shear and Torsion Interaction in Prestressed Hollow Core Slabs". Publication 05:2, 83 pp. Licentiate Thesis. Chalmers University of Technology, 2005.
- [27] Rasmus Rempling. "Constitutive Modelling of Concrete Subjected to Tensile Monotonic and Cyclic Loading". Publication 06:4, 59 pp. Licentiate Thesis. Chalmers University of Technology, 2006.
- [28] Kamyab Zandi Hanjari. "Load-Carrying Capacity of Damaged Concrete Structures". Publication 08:6, 108 pp. Licentiate Thesis. Chalmers University of Technology, 2008.
- [29] Anette Jansson. "Fibres in Reinforced Concrete Structures—analysis, experiments and design". Publication 08:3, 84 pp. Licentiate Thesis. Chalmers University of Technology, 2008.
- [30] Ulrika Nyström. "Concrete Structures Subjected to Blast and Fragment Impacts, Numerical Simulations of Reinforced and Fibre-reinforced Concrete". Publication 08:4, 117 pp. Licentiate Thesis. Chalmers University of Technology, 2008.
- [31] Hendrik Schlune. "Improved Bridge Evaluation. Finite Element Model Updating and Simplified non-linear Analysis". Publication 09:01, 20 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2009.
- [32] Filip Nilenius. "On Mesoscale Modelling of Coupled Chloride-Moisture Transport in Concrete". Publication 11:3, 10 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2011.
- [33] David Fall. "Reinforcement in Tailor-made Concrete Structures". Publication 11:5, 25 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2011.
- [34] Natalie Williams Portal. "Sustainability and Flexural Behaviour of Textile Rein-forced Concrete". Publication 2013:9, 53 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2013.
- [35] Mohammad Tahershamsi. "Anchorage of Corroded Reinforcement in Existing Concrete Structures: Experimental Study". Publication 13:10, 19 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2013.
- [36] Carlos Gil Berrocal. "Chloride Induced Corrosion of Steel Bars in Fibre Reinforced Concrete". Publication 15:01, 75 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2015.
- [37] Jiangpeng Shu. "Structural Analysis of Existing RC Bridge Deck Slabs Structural Analysis of Existing RC Bridge Deck Slabs". Publication 15:03, 17 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2015.
- [38] Jonas Ekström. "Concrete Structures Subjected to Blast Loading: Fracture due to dynamic response". Publication 2015:04, 20 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2015.

- [39] Mattias Blomfors. "Reliable Assessments of Concrete Structures with Corroded Reinforcement: An Engineering Approach". Publication 2017:06, 46 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2017.
- [40] Daniel Ekström. "Integrated design and construction for bridges: Key aspects and benefits". 22 pp. + papers. Licentiate Thesis. Chalmers University of Technology,2017.
- [41] Adam Sciegaj. "Multiscale Modelling of Reinforced Concrete". 18 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2018.
- [42] Jincheng Yang. "Flexural Strengthening of Reinforced Concrete Beams Using Externally Bonded CFRP: An Innovative Method for the Application of Prestressed CFRP Laminates". 27 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2019.
- [43] Samanta Robuschi. "Anchorage of naturally-corroded, plain bars in Reinforced Concrete structures". 38 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2019.
- [44] Alexandre Mathern. "Sustainability-, Buildability- and Performance-driven Structural Design". 39 pp. + papers. Licentiate Thesis. Chalmers University of Technology, 2019.